



Article Solvability Criteria for Uncertain Differential Equations and Their Applicability in an Economic Lot-Size Model with a Type-2 Interval Phenomenon

Mostafijur Rahaman ¹, Rakibul Haque ², Shariful Alam ¹, Sebastian Zupok ³, Soheil Salahshour ^{4,5,6}, Fariba Azizzadeh ⁷ and Sankar Prasad Mondal ^{2,*}

- ¹ Department of Mathematics, Indian Institute of Engineering Science and Technology, Shibpur, Howrah 711103, India; mostafijur.rs2019@math.iiests.ac.in (M.R.); salam@math.iiests.ac.in (S.A.)
- ² Department of Applied Mathematics, Maulana Abul Kalam Azad University of Technology, West Bengal, Haringhata, Nadia 741249, India; rakibulhaque04@gmail.com
- ³ Faculty of Social Sciences and Computer Science, Wyzsza Szkoła Biznesu National Louis University, 33-300 Nowy Sacz, Poland; szupok@wsb-nlu.edu.pl
- ⁴ Faculty of Engineering and Natural Sciences, Bahcesehir University, 34353 Istanbul, Turkey; soheil.salahshour@eng.bau.edu.tr
- ⁵ Department of Computer Science and Mathematics, Lebanese American University, Beirut P.O. Box 13-5053, Lebanon
- ⁶ Faculty of Engineering and Natural Sciences, Istanbul Okan University, 34959 Istanbul, Turkey
- ⁷ Department of Management, Islamic Azad University, Urmia 57169-6389, Iran; azizzadeh2010@yahoo.com
- * Correspondence: sankarprasad.modal@makautwb.ac.in

Abstract: Interval numbers comprise potential fields of application and describe the imprecision brought on by the flexible nature of data between boundaries. The recently added type-2 interval number allows a more thorough understanding of interval numbers. Differential equations are commonly employed in mathematical models to handle dynamic problems. It is essential to provide theories of differential equations to describe these models in an ambiguous environment controlled by type-2 interval numbers. This study proposes the type-2 interval context solvability requirements for the initial-valued first differential equation. The conditions for the solution's existence and uniqueness must be met before a brief manifestation of the solution under generalized Hukuhara differentiation occurs. An economic order quantity model analysis in a type-2 interval scenario uses a generalized Hukuhara differentiation approach.

Keywords: generalized Hukuhara difference; type-2 interval number; generalized Hukuhara differentiability; Riemann integrability; differential equation; existence and uniqueness theorem; lot-size model

1. Introduction

Almost all instances, including observations, measurements, and decision making, entail some degree of imprecision. To roughly trace the real-world process, computational simulations evolve. It necessitates a piece of mathematical equipment to address such hazy issues because the parameters and variables employed in mathematical models are subject to uncertainty. The commonly used mathematical foundation of interval numbers and fuzzy logic might meet the requirement. While the fuzzy theory offers an understanding of a sense of belongingness, interval numbers can clarify the data with ambiguity between boundaries. Calculus and algebraic topics are included in interval numbers and potential applications. Pioneering works in the interval number theory were contributed by Moore [1], Aubin and Frankowska [2], and Lakshimikantham [3]. The initial trend was about establishing the notion of interval arithmetic and its applications. In these concerns, several worthy research findings [4–10] were attached in the literature on interval uncertainty, addressing



Citation: Rahaman, M.; Haque, R.; Alam, S.; Zupok, S.; Salahshour, S.; Azizzadeh, F.; Mondal, S.P. Solvability Criteria for Uncertain Differential Equations and Their Applicability in an Economic Lot-Size Model with a Type-2 Interval Phenomenon. *Symmetry* **2023**, *15*, 1883. https://doi.org/10.3390/ sym15101883

Academic Editor: Muhammad Riaz

Received: 6 September 2023 Revised: 1 October 2023 Accepted: 3 October 2023 Published: 7 October 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the introduction of interval numbers and their applications in applied science, engineering, and management. The application area of the interval number theory was broadened in reliability optimization [11], design [12,13], and bio-mathematical models [14]. The interval number expresses the sense of uncertainty due to the variance of concerned data in a specific range. In certain instances of mathematical modeling, the upper and lower ends of the interval numbers may be flexible. For example, in the retail sector, customer demand pattern alters between ranges. That is, the ranges' ends are also elastic. To address this scenario, we must broaden the definition of uncertainty as provided by interval numbers. In this context, the philosophy of type-2 interval numbers comes into the picture, generalizing the notion of interval uncertainty. Rahman et al. [15] defined the type-2 interval numbers. Discussions of the solving criteria for differential equations in type-2 interval surroundings and their direct effects on constructing an economic lot-size model are contained in this research. This paper advances the theory of interval differential equations using the recently established approach to type-2 interval number-based uncertainty. Subsequent segments of this section, therefore, outline the literature on type-2 interval number theory, interval calculus, differential equations, and the research voids that prompted authors to develop the hypotheses in this study.

1.1. Literature on Type-2 Interval Number Theory

Interval numbers carry uncertainties due to the inexactness of the parameters involved in decision-making procedures. However, addressing ambiguity in portraying data between specific upper and lower bounds is challenging. It is better to describe the lower and upper limits of the intervals to be flexible within bounds. Rahman et al. [15] hinted at an imprecise phenomenon where the ends of such intervals again fluctuate between bounds. In that study, they also introduced some preliminaries on the limit, continuity, and differentiability of the type-2 interval numbers. Another succeeding research [16] portrayed the arithmetic properties of the interval numbers and their consequences. They also discussed an inventory model in the light of type-2 interval theory. The optimization approaches of both constrained and unconstrained types for dealing with the mathematical model under the mentioned interval uncertainty were discussed later by Rahman et al. [17]. Rahman et al. [18] also used a genetic algorithm to address the inventory model under type-2 interval uncertainty in a metaheuristic optimization approach. The real-coded self-organizing migrating genetic algorithm was utilized to discuss warehousing decision strategies in a type-2 interval scenario. Das et al. [19] dealt with ranking type-2 interval numbers and manifested the consequence on unconstrained non-linear programming models. Rahaman et al. [20] added some fundamental definitions of type-2 interval arithmetic and conformable calculus. We have yet to notice any research article emphasizing the theory of differential equations accounting for type-2 interval uncertainties.

1.2. Literature on Interval-Valued Calculus and Differential Equations

We surveyed the research on conventional interval-valued calculus and differential equations to introduce the mathematical foundations of type-2 interval-valued differential equations. Interval numbers' arithmetic characteristics are distinct from those of crisp numbers. For example, -A is not the additive inverse of the interval number A. The difference between two interval numbers is not uniquely defined. This fact has the consequence of introducing the Hukuhara and generalized Hukuhara differences. Therefore, researchers introduced integral and differential calculus theories for interval-valued functions. A remarkable contribution was made by Stefanini and Bede [21]. They discussed differentiation using the generalized Hukuhara difference for functions involving interval impreciseness. A numerical solution scheme for interval-valued differential equations under the mentioned approach was credited by an immediate investigation [22]. In this context, a comparative manifestation of fuzzy and interval-valued calculus was contributed by Stefanini [23]. The notion of generalized Hukuhara difference was one of the central concerns of the paper by Tao and Zhang [24], where they characterized the functions

incurring the interval uncertainty. Lupulescu [25] discussed the properties of the intervalvalued functions concerning integral and differential calculus. Agarwal et al. [26] was the pioneer in introducing fractional calculus for interval-valued functions. The theories of fuzzy and interval fractional calculus evolved together. In this context, we can address the works [27–33] on the ideas and applications of the Riemann–Liouville and Caputo fractional differential equations under fuzzy uncertainty. The generalized Hukuhara difference for fuzzy and interval-valued functions was used in introducing the fractional derivatives and integrals of those functions. Lupulescu [34] contributed a detailed and worthy study on the differential and integral calculus of fractional order for interval-valued functions in this directrix. The confirmable definition of derivative and integration was discussed in the interval frame by Salahshour et al. [35].

1.3. Motivations and Objectives of This Paper

The synopsis of the previous discussion guides us in the direction of the inspiration for this paper. We uncover the following details:

- 1. The theory of interval numbers and its applications has potential literature involving arithmetic properties of interval numbers, differential and integral calculus of interval-valued functions, and optimization approaches concerning interval-valued objective functions and constraints.
- 2. The standard interval number theory faces challenges when attempting to make sense of certain perplexing circumstances. The type-2 interval concept was created, broadening the interval uncertainty range.
- 3. By adding preliminary arithmetic features and optimization issues, the type-2 interval number theory is constrained. Several computational problems were solved using type-2 interval uncertainty.

By integrating the previously described ideas, we arrive at our motives and established goals for this work, which are as follows:

- 1. Several mathematical models involve differential equations. The differential equation theory under said uncertainty needs to discuss such models under type-2 interval uncertainty. Addressing the mathematical model using crisp differential equations succeeding by optimization under type-2 interval uncertainty may not be a convincing approach in many cases. Differential equation approach under generalized Hukuhara differentiability of type-2 interval-valued functions for discussing and analyzing the whole mathematical problem may be fruitful alternatives in this context. Thus, we find the usefulness of the type-2 interval-valued differential equations.
- 2. Before going for a detailed manifestation of the differential equation under type-2 interval uncertainty, we address the existence and uniqueness of the solvability criteria.
- 3. In this paper, we endeavor to determine the conditions for the first-order type-2 interval-valued differential equation's existence of a single solution. The solution of a first-order linear differential equation under various scenarios with type-2 interval uncertainty is next discussed, followed by a discussion of the existence and uniqueness theory. The proposed approach is followed promptly by an appraisal of an economic order quantity model.

1.4. Summary of the Organization of This Paper

Following the introduction section, this paper is organized as follows. Section 2 discusses some preliminaries regarding the theory of type-2 interval numbers. Section 3 discusses the conditions for the existence and uniqueness of a differential equation under said uncertainty. Section 4 discusses cases for solving a first-order linear differential equation under different type-2 interval uncertainty cases. Section 5 describes an EOQ model for deteriorated commodities using the generalized Hukuhara differentiation approach for type-2 interval-valued functions. Section 6 concludes this paper.

2. Mathematical Preliminaries

This section revisits some mathematical preliminaries on type-2 interval uncertainty based on the proposed theory introduced in this paper.

Definition 1 ([15,16]). An interval number given by $\mathcal{A} = [[a_{LL}, a_{LU}], [a_{UL}, a_{UU}]]$ in which $a_{LL} \leq a_{LU} \leq a_{UL} \leq a_{UU}$ is called a T2IN.

A T2IN has uncertain upper and lower bounds. The lower and upper bounds lie in the intervals $[a_{LL}, a_{LU}]$, and $[a_{UL}, a_{UU}]$, respectively. As a particular case, if the upper and lower bounds lie in $\{a_{LL}, a_{LU}\}$ and $\{a_{UL}, a_{UU}\}$, respectively, then a T2IN includes the uncertain combinations of four interval numbers, namely, $[a_{LL}, a_{UL}], [a_{LL}, a_{UU}], [a_{LL}, a_{UL}], [a_{LL}, a_{UL}], and <math>[a_{LU}, a_{UU}]$. Therefore, a T2IN is an uncertain combination of interval numbers. Thus, a T2IN generalizes the traditional interval numbers. Figure 1 depicts the sense of generalization in interval uncertainty made by a T2IN.



Figure 1. Graphical representation of a T2IN is given as $\mathcal{A} = [[a_{LL}, a_{LU}], [a_{UL}, a_{UU}]]$.

Definition 2 ([15]). Let $\mathcal{A} = [[a_{LL}, a_{LU}], [a_{UL}, a_{UU}]]$ and $\mathcal{B} = [[b_{LL}, b_{LU}], [b_{UL}, b_{UU}]]$ be two *T2INs and* γ *be any real number. Then, addition and scalar multiplication are given as*

 $\begin{aligned} Addition. \ \mathcal{A} \boxplus \mathcal{B} &= [[a_{LL} + b_{LL}, a_{LU} + b_{LU}], [a_{UL} + b_{UL}, a_{UU} + b_{UU}]]\\ Scalarmultiplication. \ \gamma \boxdot \mathcal{A} &= \begin{cases} [[\gamma a_{LL}, \gamma a_{LU}], [\gamma a_{UL}, \gamma a_{UU}]] \ for \ \gamma \geq 0\\ [[\gamma a_{UU}, \gamma a_{UL}], [\gamma a_{LU}, \gamma a_{LL}]], \ for \ \gamma < 0 \end{cases} \end{aligned}$

Definition 3 ([19]). The interval width of $\mathcal{A} = [[a_{LL}, a_{LU}], [a_{UL}, a_{UU}]]$ is defined as $\mathcal{W}(\mathcal{A}) = [a_{UL} - a_{LU}, a_{UU} - a_{LL}]$.

For example, let a T2IN be taken as $\mathcal{A} = [[2,3], [10,11]]$. Then, the interval width is $(\mathcal{A}) = [7, 9]$, an interval itself. A T2IN is a degenerate interval number when the bounds were crisp instead of intervals. Then, $a_{LL} = a_{LU} = a_L$, $a_{UL} = a_{UU} = a_U$, and $\mathcal{W}(\mathcal{A}) = a_U - a_L$ is the width of the interval number $[a_L, a_U]$.

Definition 4 ([19]). Let $\mathcal{A} = [[a_{LL}, a_{LU}], [a_{UL}, a_{UU}]]$ and $\mathcal{B} = [[b_{LL}, b_{LU}], [b_{UL}, b_{UU}]]$ be two T2INs. Then, $\mathcal{W}(\mathcal{A}) \succeq_{in} \mathcal{W}(\mathcal{B})$ is equivalent to $a_{UL} - a_{LU} \ge b_{UL} - b_{UL}$ and $a_{UU} - a_{LL} \ge b_{UU} - b_{LL}$. In that case, we say that \mathcal{A} has a greater interval width than \mathcal{B} .

Definition 5 ([15]). *The generalized Hukuhara difference between two T2Ins* $\mathcal{A} = [[a_{LL}, a_{LU}], [a_{UL}, a_{UU}]]$ and $\mathcal{B} = [[b_{LL}, b_{LU}], [b_{UL}, b_{UU}]]$ is defined as

$$\mathcal{A} \boxminus_{g} \mathcal{B} = \frac{[[\min\{a_{LL} - b_{LL}, a_{UU} - b_{UU}\}, \min\{a_{LU} - b_{LU}, a_{UL} - b_{UL}\}]}{[\max\{a_{LL} - b_{LL}, a_{UU} - b_{UU}\}, \max\{a_{LU} - b_{LU}, a_{UL} - b_{UL}\}]]}$$

Lemma 1 ([19]). Let $\mathcal{A} = [[a_{LL}, a_{LU}], [a_{UL}, a_{UU}]]$ and $\mathcal{B} = [[b_{LL}, b_{LU}], [b_{UL}, b_{UU}]]$ be two T2INs. Then, the generalized Hukuhara difference between them depends on interval widths as follows:

$$\mathcal{A} \boxminus_{g} \mathcal{B} = \begin{cases} [[a_{LL} - b_{LL}, a_{LU} - b_{LU}], [a_{UL} - b_{UL}, a_{UU} - b_{UU}]], \text{ when } \mathcal{W}(\mathcal{A}) \succcurlyeq_{in} \mathcal{W}(\mathcal{B}) \\ [[a_{UU} - b_{UU}, a_{UL} - b_{UL}], [a_{LU} - b_{LU}, a_{LL} - b_{LL}]], \text{ when } \mathcal{W}(\mathcal{B}) \succcurlyeq_{in} \mathcal{W}(\mathcal{A}) \end{cases}$$

3. Existence and Uniqueness of Initial Valued Differential Equations

A type-2 interval number $C = [[c_{LL}, c_{LU}], [c_{UL}, c_{UU}]]$ will be the generalized Hukuhara difference of \mathcal{A} and \mathcal{B} only if either of the following conditions is satisfied: $\mathcal{A} = \mathcal{B} \boxplus C$ when $\mathcal{W}(\mathcal{A}) \succeq_{in} \mathcal{W}(\mathcal{B})$, or $\mathcal{B} = \mathcal{A} \boxplus \{(-1) \boxdot C\}$ when $\mathcal{W}(\mathcal{B}) \succeq_{in} \mathcal{W}(\mathcal{A})$.

A function $\|.\|_4 : K_4 \to R^+$ defined by $\|\mathcal{A}\|_4 = max\{|a_{LL}|, |a_{LU}|, |a_{UL}|, |b_{UU}|\}$ is the norm on the set of all type-2 interval numbers. The norm $\|.\|_4$ gives the metric Δ_4 : $K_4 \times K_4 \to R^+$ defined by $\Delta_4(\mathcal{A}, \mathcal{B}) = max\{|a_{LL} - b_{LL}|, |a_{LU} - b_{LU}|, |a_{UL} - b_{UL}|, |a_{UL} - b$

Definition 6. A function \mathcal{F} defined as $\mathcal{F}(x) = [[f_{LL}(x), f_{LU}(x)], [f_{UL}(x), f_{UU}(x)]]$, where each of f_{LL}, f_{LU}, f_{UL} , and f_{UU} are all crisp functions maintaining the order $f_{LL}(x) \le f_{LU}(x) \le f_{UL}(x) \le f_{UL}(x) \le f_{UL}(x) \le f_{UU}(x)$ throughout the domain of definitions \mathcal{D} is called a T2IF on \mathcal{D} .

 $C([a, b], K_4)$, the set of all type-2 interval-valued continuous functions defined on ([a, b]), is a closed subset of $B([a, b], K_4)$, and hence it is a complete norm space.

Theorem 1. Let \mathcal{F} be a T2IF defined on [a,b] that is given by $\mathcal{F}(x) = [[f_{LL}(x), f_{LU}(x)], [f_{UL}(x), f_{UU}(x)]]$. If \mathcal{F} is generalized Hukuhara differentiable and \mathcal{W} -monotonic on [a,b], then each of f_{LL} , f_{LU} , f_{UL} , f_{UU} are differentiable on [a,b], and furthermore:

(i)
$$\frac{d\mathcal{F}(x)}{dx} = \left[\left[\frac{df_{LL}(x)}{dx}, \frac{df_{LU}(x)}{dx} \right], \left[\frac{df_{UL}(x)}{dx}, \frac{df_{UU}(x)}{dx} \right] \right], \text{ when } \mathcal{F} \text{ is } \mathcal{W}\text{-increasing for all } x \in [a, b].$$

(ii)
$$\frac{d\mathcal{F}(x)}{dx} = \left[\left[\frac{df_{UU}(x)}{dx}, \frac{df_{UL}(x)}{dx} \right], \left[\frac{df_{UU}(x)}{dx}, \frac{df_{LL}(x)}{dx} \right] \right], \text{ when } \mathcal{F} \text{ is } \mathcal{W}\text{-decreasing for all } x \in [a, b].$$

Proof of Theorem 1. Let $x \in [a, b]$ arbitrary and, \mathcal{F} is generalized Hukuhara differentiable. That is, $\frac{d\mathcal{F}(x)}{dx} = \lim_{\epsilon \to 0} \frac{\mathcal{F}(x_0 + \epsilon) \boxminus_g \mathcal{F}(x_0)}{\epsilon}$ exists finitely, and \mathcal{F} is \mathcal{W} -monotone on [a, b]. We consider two cases as follows:

Case 1. When W-increasing on [a, b]

Since $\frac{d\mathcal{F}(x)}{dx} = \lim_{\epsilon \to 0} \frac{\mathcal{F}(x+\epsilon) \boxminus_g \mathcal{F}(x)}{\epsilon}$ exists, then $\lim_{\epsilon \to 0^-} \frac{\mathcal{F}(x+\epsilon) \boxminus_g \mathcal{F}(x)}{\epsilon} = \lim_{\epsilon \to 0^+} \frac{\mathcal{F}(x+\epsilon) \boxminus_g \mathcal{F}(x)}{\epsilon}$. Then, by Lemma 1, we have obtained the type-2 interval equation as follows:

$$\lim_{\epsilon \to 0^{-}} \frac{\begin{bmatrix} [f_{UU}(x+\epsilon) - f_{UU}(x), f_{UL}(x+\epsilon) - f_{UL}(x)] \\ [f_{LU}(x+\epsilon) - f_{LU}(x), f_{LL}(x+\epsilon) - f_{LL}(x)] \\ \epsilon \end{bmatrix}}{\epsilon} = \lim_{\epsilon \to 0^{+}} \frac{\begin{bmatrix} [f_{LL}(x+\epsilon) - f_{LL}(x), f_{LU}(x+\epsilon) - f_{LU}(x)] \\ [f_{UL}(x+\epsilon) - f_{UL}(x), f_{UU}(x+\epsilon) - f_{UU}(x)] \\ \epsilon \end{bmatrix}}{\epsilon}$$

$$\begin{bmatrix} \lim_{\epsilon \to 0^{-}} \frac{f_{LL}(x+\epsilon) - f_{LL}(x)}{\epsilon}, \lim_{\epsilon \to 0^{-}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{-}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to 0^{+}}, \lim_{\epsilon \to 0^{+}} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon} \end{bmatrix}_{\epsilon \to$$

Equating components of T2IN, we conclude that each of f_{LL} , f_{LU} , f_{UL} , f_{UU} , f_{UU} are differentiable on $x \in [a, b]$, and in that case, $\frac{d\mathcal{F}(x)}{dx} = \left[\left[\frac{df_{LL}(x)}{dx}, \frac{df_{LU}(x)}{dx} \right], \left[\frac{df_{UL}(x)}{dx}, \frac{df_{UU}(x)}{dx} \right] \right]$.

Case 2. When W-decreasing on [a, b]

Since $\frac{d\mathcal{F}(x)}{dx} = \lim_{\epsilon \to 0} \frac{\mathcal{F}(x+\epsilon) \boxminus_g \mathcal{F}(x)}{\epsilon}$ exists, then $\lim_{\epsilon \to 0^-} \frac{\mathcal{F}(x+\epsilon) \boxminus_g \mathcal{F}(x)}{\epsilon} = \lim_{\epsilon \to 0^+} \frac{\mathcal{F}(x+\epsilon) \boxminus_g \mathcal{F}(x)}{\epsilon}$. Then, by Lemma 1, we have obtained the type-2 interval equation as follows:

$$\lim_{\epsilon \to 0^{-}} \frac{\left[\begin{bmatrix} f_{LL}(x+\epsilon) - f_{LL}(x), f_{LU}(x+\epsilon) - f_{LU}(x) \end{bmatrix}}{[f_{UL}(x+\epsilon) - f_{UL}(x), f_{UU}(x+\epsilon) - f_{UU}(x)]' \right]}{\epsilon} = \lim_{\epsilon \to 0^{+}} \frac{\left[\begin{bmatrix} f_{UU}(x+\epsilon) - f_{UU}(x), f_{UL}(x+\epsilon) - f_{UL}(x) \end{bmatrix}}{[f_{LU}(x+\epsilon) - f_{LU}(x), f_{LL}(x+\epsilon) - f_{LL}(x)]' \right]}{\epsilon}$$

The above equation leads to the consequence as follows:

$$\begin{bmatrix} \lim_{\epsilon \to 0^-} \frac{f_{UU}(x+\epsilon) - f_{UU}(x)}{\epsilon}, \lim_{\epsilon \to 0^-} \frac{f_{UL}(x+\epsilon) - f_{UL}(x)}{\epsilon} \\ \lim_{\epsilon \to 0^-} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon}, \lim_{\epsilon \to 0^-} \frac{f_{LL}(x+\epsilon) - f_{LL}(x)}{\epsilon} \end{bmatrix}, = \begin{bmatrix} \lim_{\epsilon \to 0^+} \frac{f_{UU}(x+\epsilon) - f_{UU}(x)}{\epsilon}, \lim_{\epsilon \to 0^+} \frac{f_{UL}(x+\epsilon) - f_{LL}(x)}{\epsilon} \\ \lim_{\epsilon \to 0^+} \frac{f_{LU}(x+\epsilon) - f_{LU}(x)}{\epsilon}, \lim_{\epsilon \to 0^+} \frac{f_{LL}(x+\epsilon) - f_{LL}(x)}{\epsilon} \end{bmatrix}, \end{bmatrix}$$

Equating components of T2IN, we conclude that each of f_{LL} , f_{LU} , f_{UL} , f_{UU} are differentiable on $x \in [a, b]$, and in that case, $\frac{d\mathcal{F}(x)}{dx} = \left[\left[\frac{df_{UU}(x)}{dx}, \frac{df_{UL}(x)}{dx} \right], \left[\frac{df_{LU}(x)}{dx}, \frac{df_{LL}(x)}{dx} \right] \right]$. \Box

Definition 7. Let \mathcal{F} be a T2IF defined on [a,b] given by $\mathcal{F}(x) = [[f_{LL}(x), f_{LU}(x)], [f_{UL}(x), f_{UU}(x)]]$. Then, \mathcal{F} is Riemann integrable on [a,b] when f_{LL} , f_{LU} , f_{UL} , and f_{UU} are all Riemann integrable functions on [a,b]. In that case, we define the Riemann integration of the T2IF \mathcal{F} as follows:

$$\int_{a}^{b} \mathcal{F}(x) dx = \left[\left[\int_{a}^{b} f_{LL}(x) dx, \int_{a}^{b} f_{LU}(x) dx \right], \left[\int_{a}^{b} f_{UL}(x) dx, \int_{a}^{b} f_{UU}(x) dx \right] \right]$$

Now, we introduce the existence and uniqueness of solvability of the differential equation in a type-2 interval environment.

Let us consider a type-2 interval-valued differential equation of first order given as

$$\begin{cases} \frac{d\widetilde{y_2}}{dx} = \widetilde{F_2}(x, \ \widetilde{y_2}(x)) \\ \widetilde{y_2}(x_0) = \widetilde{y_{02}} \end{cases}$$
(1)

In Equation (1), \tilde{y}_2 is a type-2 valued dependent variable while *x* is a crisp independent variable.

In the system given by Equation (1), $F_2 : [a, b] \times I_2 \rightarrow I_2$ is a type-2 interval-valued function (where I_2 is the set of all type-2 interval numbers), which is given by four components as follows:

$$\widetilde{\mathbf{F}_2}(x,\,\widetilde{y_2}(x)) = \left[\left[f_{LL}(x,\,\widetilde{y_2}(x)),\,f_{LU}(x,\,\widetilde{y_2}(x)) \right], \left[f_{UL}(x,\,\widetilde{y_2}(x)),\,f_{UU}(x,\,\widetilde{y_2}(x)) \right] \right]$$

Also, the dependent variable $\widetilde{y_2}$ and initial value $\widetilde{y_{02}}$ are given respectively as $\widetilde{y_2}(x) = \left[\left[y_{LL}(x, \widetilde{y_2}), y_{LU}(x, \widetilde{y_2}) \right], \left[y_{UL}(x, \widetilde{y_2}), y_{UU}(x, \widetilde{y_2}) \right] \right]$ and $\widetilde{y_{02}} = \left[\left[y_{0LL}, y_{0LU} \right], \left[y_{0UL}, y_{0UU} \right] \right]$. In

Equation (1), $\frac{dy_2}{dx}$ represents the generalized Hukuhara differentiability of the type-2 valued dependent variable \tilde{y}_2 concerning *x*, the crisp independent variable.

Now, Equation (1) is equivalent to the integral equation

$$\widetilde{y_2}(x) \boxminus_{gH} \widetilde{y_{02}} = \int_{x_0}^x \widetilde{F_2}(u, \, \widetilde{y_2}(u)) du.$$
⁽²⁾

Using the sense of the generalized Hukuhara difference of a type-2 interval-valued function, Equation (2) is equivalent to either

$$\widetilde{y}_2(x) = \widetilde{y}_{02} \boxplus \int_{x_0}^x \widetilde{F}_2(u, \widetilde{y}_2(u)) du$$
(3)

Or,
$$\widetilde{y_{02}} = \widetilde{y_2}(x) \boxplus (-1) \odot \int_{x_0}^x \widetilde{F_2}(u, \widetilde{y_2}(u)) du$$
 (4)

Theorem 2. We define a closed ball in type-2 interval-valued functions as follows:

$$\overline{B_r}(\widetilde{y_{02}}) = \left\{ \widetilde{z_2} \in I_2 : \mu_2(\widetilde{z_2}, \widetilde{y_{02}}) \le r \right\}$$

where μ_2 is the type-2 interval numbers metric and \mathbf{r} is a positive real number. In the above definition, $\overline{B_r}(\widetilde{y_{02}})$ is the closed ball having a radius \mathbf{r} and a center at $\widetilde{y_{02}}$. Then, we construct a region \mathbb{R}^* as follows: $\mathbb{R}^* = [x_0, x_0 + h] \times \overline{B_r}(\widetilde{y_{02}})$, where x_0, h are crisp numbers and $\widetilde{y_{02}}$ is a type-2 interval number. Then, we construct $\widetilde{F_2} : \mathbb{R}^* \to I_2$ as a continuous, non-trivial type-2 valued function, which satisfies the Lipschitz condition as follows:

$$\mu_2(\widetilde{\mathbf{F}}_2(x,\widetilde{y}_2(x)),\widetilde{\mathbf{F}}_2(x,\widetilde{z}_2(x))) \le k\mu_2(\widetilde{y}_2(x),\widetilde{z}_2(x))$$
(5)

for all $(x, \tilde{y_2}(x))$, $(x, \tilde{z_2}(x))$ in \mathbb{R}^* .

Then, the system given by Equation (1) has two unique solutions, \tilde{y}_2^{-1} and \tilde{y}_2^{-2} , provided by the successive iterations as follows:

$$\begin{cases} \overset{\sim}{y_{(n+1),2}}^{-1}(x) = \overset{\sim}{y_{02}} \boxplus \int_{x_0}^{x} \overset{\sim}{F_2}(u, y_{n,2}^{-1}(u)) du \\ \overset{\sim}{y_{02}}^{-2} = \overset{2}{y_{(n+1),2}}^{-2}(x) \boxplus (-1) \odot \int_{x_0}^{x} \overset{\sim}{F_2}(u, y_{n,2}^{-2}(u)) du \end{cases}$$
(6)

Proof of Theorem 2. It is perceived that Equation (1) is equivalent to either Equation (3) or Equation (4).

Since I_2 , the space of all type-2 interval numbers form a metric space under the metric μ_2 , (I_2, μ_2) is a metric Hausdroff space and is therefore locally compact. Therefore, $\overline{B_r}(\widetilde{y_{02}})$ is compact. Then, $f_{UL} - f_{LU}$ and $f_{UU} - f_{LL}$ are bounded functions defined on R^* . Thus, positive real numbers M_1 and M_2 exist such that the following are true:

$$\begin{cases} (f_{UL} - f_{LU})(x, \widetilde{y_2}) \le M_1 \\ (f_{UU} - f_{LL})(x, \widetilde{y_2}) \le M_2 \end{cases}$$

$$\tag{7}$$

Therefore, the interval width of $\overset{\sim}{F_2}$ satisfies the following inequalities:

$$\mathcal{W}(F_2(x, \widetilde{y_2}(x))) \leq_{in} [M_1, M_2]$$
(8)

Since $\widetilde{y_{02}}$ is a non-trivial type-2 interval number, $\mathcal{W}(\widetilde{y_{02}}) \succeq_{in} [0,0]$. That is, $(y_{0UL} - y_{0LU}) \ge 0$ and $(y_{0UU} - y_{0LL}) \ge 0$. Let $h_1 = min\left\{h, \frac{(y_{0UL} - y_{0LU})}{M_1}, \frac{(y_{0UU} - y_{0LL})}{M_2}\right\}$. Now, negation of $\int_{x_0}^x \widetilde{F_2}(u, \widetilde{y_2}(u)) du$ can be obtained as follows:

$$(-1) \odot \int_{x_0}^x F_2(u, \widetilde{y_2}(u)) du = \left[\left[-\int_{x_0}^x f_{UU}(u, \widetilde{y_2}(u)) du, -\int_{x_0}^x f_{UL}(u, \widetilde{y_2}(u)) du \right], \left[-\int_{x_0}^x f_{LU}(u, \widetilde{y_2}(u)) du, -\int_{x_0}^x f_{LL}(u, \widetilde{y_2}(u)) du \right] \right]$$

Therefore, the interval width representation of the above equation can be obtained as follows:

$$\mathcal{W}\left((-1)\odot\int_{x_0}^x \widetilde{F_2}(u, \widetilde{y_2}(u))du\right) = \left[\int_{x_0}^x (f_{UU} - f_{LL})\left(u, \widetilde{y_2}(u)\right)du, \int_{x_0}^x (f_{UL} - f_{LU})\left(u, \widetilde{y_2}(u)\right)du\right]$$

$$\leq_{in} [M_1, M_2](x - x_0) \leq_{in} [M_1h_1, M_2h_1] \leq_{in} [(y_{0UL} - y_{0LU}), (y_{0UU} - y_{0LL})] = \mathcal{W}(\widetilde{y_{02}})$$

That is,
$$\mathcal{W}\left((-1)\odot\int_{x_0}^x\widetilde{F}_2(u,\widetilde{y}_2(u))du\right)\leq_{in}\mathcal{W}(\widetilde{y}_{02})$$
 (9)

Therefore, the Hukuhara difference $y_{02}^{\sim} \boxminus_H (-1) \odot \int_{x_0}^x \widetilde{F_2}(u, \widetilde{y_2}(u)) du$ exists.

We construct $C_0 = [x_0, x_0 + h_1] \times \overline{B_r}(y_{02})$, a compact set. Then, we consider two operations, Φ and Ψ , as follows:

$$\begin{cases} \Phi(\widetilde{y}_{2}(x_{0})) = \Psi(\widetilde{y}_{2}(x_{0})) = \widetilde{y}_{02}^{2} \\ \Phi(\widetilde{y}_{2}(x)) = \widetilde{y}_{02} \boxplus \int_{x_{0}}^{x} \widetilde{F}_{2}(u, \widetilde{y}_{2}(u)) du \\ \Psi(\widetilde{y}_{2}(x)) = \widetilde{y}_{02} \boxminus_{H} (-1) \odot \int_{x_{0}}^{x} \widetilde{F}_{2}(u, \widetilde{y}_{2}(u)) du \end{cases}$$
(10)

In System (10), the operator Φ is well-defined irrespective of the choice of the neighborhood of x_0 . But, Ψ is well-defined depending on the choice of the neighborhood of x_0 . Now, by the metric space property, we have the following results:

$$\mu_{2}\left(\Phi\left(\widetilde{y_{2}}(x)\right), \widetilde{y_{02}}\right) = \mu_{2}\left(\widetilde{y_{02}} \boxplus \int_{x_{0}}^{x} \widetilde{F_{2}}\left(u, \widetilde{y_{2}}(u)\right) du, \widetilde{y_{02}}\right)$$
$$= \mu_{2}\left(\int_{x_{0}}^{x} \widetilde{F_{2}}(u, \widetilde{y_{2}}(u)) du, 0\right) \leq \int_{x_{0}}^{x} \mu_{2}\left(\widetilde{F_{2}}(u, \widetilde{y_{2}}(u)), 0\right) du \leq M_{3}(x - x_{0})$$
(11)

The above result is obtained as $\widetilde{F_2}$, which is a continuous function on the compact set C_0 . Again,

$$\mu_{2}\left(\Psi\left(\widetilde{y_{2}}(x)\right), \widetilde{y_{02}}\right) = \mu_{2}\left(\widetilde{y_{02}} \boxminus_{H}(-1) \odot \int_{x_{0}}^{x} \widetilde{F_{2}}\left(u, \widetilde{y_{2}}(u)\right) du, \widetilde{y_{02}}\right)$$
$$= \mu_{2}\left(\int_{x_{0}}^{x} \widetilde{F_{2}}(u, \widetilde{y_{2}}(u)) du, 0\right) \le \int_{x_{0}}^{x} \mu_{2}\left(\widetilde{F_{2}}(u, \widetilde{y_{2}}(u)), 0\right) du \le M_{3}(x - x_{0})$$
(12)

Next, we consider $h_2 = min\{min\{h, \frac{(y_{0UL}-y_{0LU})}{M_1}, \frac{(y_{0UU}-y_{0LL})}{M_2}\}, \frac{r}{M_3}\}$, and we construct $C_1 = [x_0, x_0 + h_2] \times \overline{B_r(y_{02})}$. We consider all the continuous functions from C_1 to $\overline{B_r(y_{02})}$. All such functions are complete under the sup metric for type-2 interval-valued functions.

Let \tilde{y}_2 and \tilde{z}_2 be two such continuous functions defined on C_1 . Then, the following inequalities are obtained:

Then,
$$\mu_2\left(\Phi\left(\widetilde{y}_2(x)\right), \Phi\left(\widetilde{z}_2(x)\right)\right)$$

$$= \mu_2 \left(\widetilde{y_{02}} \boxplus \int_{x_0}^x \widetilde{F_2}(u, \widetilde{y_2}(u)) du, \widetilde{y_{02}} \boxplus \int_{x_0}^x \widetilde{F_2}(u, \widetilde{z_2}(u)) du \right)$$

$$= \mu_2 \left(\int_{x_0}^x \widetilde{F_2}(u, \widetilde{y_2}(u)) du, \int_{x_0}^x \widetilde{F_2}(u, \widetilde{z_2}(u)) du \right)$$

$$\leq \int_{x_0}^x \mu_2 (\widetilde{F_2}(u, \widetilde{y_2}(u))), \widetilde{F_2}(u, \widetilde{z_2}(u)) du$$

$$\leq k \int_{x_0}^x \max_{u \in [x_0, x_0 + h_2]} \mu_2 (\widetilde{y_2}(u), \widetilde{z_2}(u)) du$$

$$\leq k (x - x_0) \mu_{2, sup} (\widetilde{y_2}, \widetilde{z_2})$$

And similarly, $\mu_2\left(\Psi\left(\tilde{y_2}(x)\right), \Psi\left(\tilde{z_2}(x)\right)\right) \le k(x-x_0)\mu_{2,sup}(\tilde{y_2}, \tilde{z_2})$. If we take, $h_3 < min\{min\{h, \frac{(y_{0UL}-y_{0LU})}{M_1}, \frac{(y_{0UU}-y_{0LL})}{M_2}\}, \frac{r}{M_3}\}, \frac{1}{k}\}$, then $k(x-x_0) < 1$. Therefore, both Φ and Ψ become contractions. Since Φ is a contraction, by the Banach fixed-point theorem, Φ has a unique fixed point $y_2^{\sim 1}$, which can be obtained by Picard's successive iterations on the process Φ and, hence,

$$y_{(n+1), 2}^{\sim 1}(x) = y_{02}^{\sim} \boxplus \int_{x_0}^x \widetilde{F_2}(u, y_{n, 2}^{\sim 1}(u)) du$$

Also, Ψ is a contraction; thus, by the Banach fixed-point theorem, Ψ has a unique fixed point \tilde{y}_2^2 , which can be obtained by Picard's successive iterations on the process Ψ and, hence,

$$\widetilde{y_{(n+1),2}}^{(n+1)}(x) = \widetilde{y_{02}} \boxminus_H (-1) \odot \int_{x_0}^{x} \widetilde{F_2}(u, \widetilde{y_2}(u)) du$$

Or equivalently,

$$\tilde{y_{02}} = y_{(n+1), 2}^{2}(x) \boxplus (-1) \odot \int_{x_{0}}^{x} \tilde{F}_{2}(u, y_{n, 2}^{2}(u)) du$$

This completes the proof with the conclusion that Equation (1) has two unique solutions, which can be uniquely obtained by Equation (6), provided certain conditions hold. \Box

Remark 1. This section ends with the proof of the theorem addressing the existence and uniqueness criteria for solving the differential equation in a type-2 interval environment. The succeeding section thus provides a detailed solution approach in different cases.

4. Solutions of First-Order Linear Differential Equations in Type-2 **Interval Environment**

A first-order linear non-homogenous differential equation in a type-2 interval environment can be represented as follows:

$$\begin{cases} \frac{d\widetilde{y_2}}{dx} = \widetilde{k_2} \odot \widetilde{y_2} \boxplus \widetilde{M_2} \\ \widetilde{y_2}(0) = \widetilde{y_{02}} \end{cases}$$
(13)

The preceding text ensures the existence of the solution. Here, we consider the following two cases depending upon the generalized Hukuhara differentiability of the type-2 interval-valued dependent variable \tilde{y}_2 .

Case 1. When y_2 is generalized Hukuhara differentiable of the first type

In this case, we consider four subcases concerning the signs of the type-2 interval numbers $\tilde{k_2}$ and $\tilde{M_2}$.

Subcase 1.1. When both k_2 and M_2 are positive type-2 interval numbers

In this scenario, the type-2 interval-valued differential equation in System (13) can be replaced by the equation incorporating four components as follows:

$$\left[\left[\frac{dy_{LL}}{dx}, \frac{dy_{LU}}{dx}\right], \left[\frac{dy_{UL}}{dx}, \frac{dy_{UU}}{dx}\right]\right] = \left[\left[k_{LL}y_{LL} + M_{LL}, k_{LU}y_{LU} + M_{LU}\right], \left[k_{UL}y_{UL} + M_{UL}, k_{UU}y_{UU} + M_{UU}\right]\right]$$

This gives four differential equations, which are given below.

$$\begin{cases} \frac{dy_{LL}}{dx} = k_{LL}y_{LL} + M_{LL} \\ \frac{dy_{UL}}{dx} = k_{LU}y_{LU} + M_{LU} \\ \frac{dy_{UL}}{dx} = k_{UL}y_{UL} + M_{UL} \\ \frac{dy_{UU}}{dx} = k_{UU}y_{UU} + M_{UU} \end{cases}$$
(14)

By solving the first equation in System (14), we obtain

$$y_{LL}e^{-k_{LL}x} = -\frac{M_{LL}}{k_{LL}}e^{-k_{LL}x} + c_1$$

The initial condition $\tilde{y}_2(0) = \tilde{y}_{02}$ compromises the crisp component $y_{LL}(0) = y_{0LL}$, which provides the value of the integrating constant as $c_1 = y_{0LL} + \frac{M_{LL}}{k_{LL}}$. Hence, the first component of the type-2 interval-valued solution is obtained as follows:

$$y_{LL} = \frac{M_{LL}}{k_{LL}} \left(e^{k_{LL}x} - 1 \right) + y_{0LL} e^{k_{LL}x}$$

Similarly, the remaining components of the type-2 interval-valued solution are obtained as follows:

$$y_{LU} = \frac{M_{LU}}{k_{LU}} \left(e^{k_{LU}x} - 1 \right) + y_{0LU} e^{k_{LU}x}$$
$$y_{UL} = \frac{M_{UL}}{k_{UL}} \left(e^{k_{UL}x} - 1 \right) + y_{0UL} e^{k_{UL}x}$$
$$y_{UU} = \frac{M_{UU}}{k_{UU}} \left(e^{k_{UU}x} - 1 \right) + y_{0UU} e^{k_{UU}x}$$

Subcase 1.2. When k_2 and M_2 are positive and negative type-2 interval numbers, respectively In this scenario, the type-2 interval-valued differential equation in System (13) can be replaced by the equation incorporating four components as follows:

$$\left[\left[\frac{dy_{LL}}{dx}, \frac{dy_{LU}}{dx} \right], \left[\frac{dy_{UL}}{dx}, \frac{dy_{UU}}{dx} \right] \right] = \left[[k_{LL}y_{LL} - M_{UU}, k_{LU}y_{LU} - M_{UL}], [k_{UL}y_{UL} - M_{LU}, k_{UU}y_{UU} - M_{LL}] \right]$$
This gives four differential equations, which are given below.

This gives four differential equations, which are given below,

$$\begin{cases} \frac{dy_{LL}}{dx} = k_{LL}y_{LL} - M_{UU} \\ \frac{dy_{LU}}{dx} = k_{LU}y_{LU} - M_{UL} \\ \frac{dy_{UL}}{dx} = k_{UL}y_{UL} - M_{LU} \\ \frac{dy_{UU}}{dx} = k_{UL}y_{UL} - M_{LL} \end{cases}$$
(15)

Following the process detailed in the preceding Subcase 1.1, the four components of the solution are obtained as follows:

$$y_{LL} = \frac{M_{UU}}{k_{LL}} \left(1 - e^{k_{LL}x}\right) + y_{0LL}e^{k_{LL}x}$$
$$y_{LU} = \frac{M_{UL}}{k_{LU}} \left(1 - e^{k_{LU}x}\right) + y_{0LU}e^{k_{LU}x}$$
$$y_{UL} = \frac{M_{LU}}{k_{UL}} \left(1 - e^{k_{UL}x}\right) + y_{0UL}e^{k_{UL}x}$$
$$y_{UU} = \frac{M_{LL}}{k_{UU}} \left(1 - e^{k_{UU}x}\right) + y_{0UU}e^{k_{UU}x}$$

Subcase 1.3. When $\tilde{k_2}$ and $\tilde{M_2}$ are negative and positive type-2 interval numbers, respectively In this scenario, the type-2 interval-valued differential equation in System (13) can be replaced by the equation incorporating four components as follows:

$$\left[\left[\frac{dy_{LL}}{dx}, \frac{dy_{LU}}{dx}\right], \left[\frac{dy_{UL}}{dx}, \frac{dy_{UU}}{dx}\right]\right] = \left[\left[-k_{UU}y_{UU} + M_{LL}, -k_{UL}y_{UL} + M_{LU}\right], \left[-k_{LU}y_{LU} + M_{UL}, -k_{LL}y_{LL} + M_{UU}\right]\right]$$

This gives four differential equations, which are given below,

$$\frac{dy_{LL}}{dx} = -k_{UU}y_{UU} + M_{LL} \tag{16}$$

$$\frac{dy_{LU}}{dx} = -k_{UL}y_{UL} + M_{LU} \tag{17}$$

$$\frac{dy_{UL}}{dx} = -k_{LU}y_{LU} + M_{UL} \tag{18}$$

$$\frac{dy_{UU}}{dx} = -k_{LL}y_{LL} + M_{UU} \tag{19}$$

From Equations (16) and (19) of the above system, we obtain

$$\frac{d^2 y_{LL}}{dx^2} = -k_{UU} \frac{dy_{UU}}{dx}$$
$$\frac{d^2 y_{LL}}{dx^2} = -k_{UU} [-k_{LL} y_{LL} + M_{UU}]$$
$$\frac{d^2 y_{LL}}{dx^2} - k_{LL} k_{UU} y_{LL} = -k_{UU} M_{UU}$$

Therefore, the solution is

$$y_{LL}(x) = c_3 e^{\sqrt{k_{LL}k_{UU}}x} + c_4 e^{-\sqrt{k_{LL}k_{UU}}x} + \frac{1}{D^2 - k_{LL}k_{UU}}(-k_{UU}M_{UU})$$
$$y_{LL}(x) = c_3 e^{\sqrt{k_{LL}k_{UU}}x} + c_4 e^{-\sqrt{k_{LL}k_{UU}}x} + \frac{M_{UU}}{k_{LL}}$$
(20)

Using Equation (20) in Equation (16) and find the value of $y_{UU}(x)$ as follows:

$$y_{UU}(x) = \frac{1}{k_{UU}} \left[M_{LL} - \sqrt{k_{LL}k_{UU}} \left(c_3 e^{\sqrt{k_{LL}k_{UU}}x} - c_4 e^{-\sqrt{k_{LL}k_{UU}}x} \right) \right]$$
(21)

Using the initial conditions $y_{LL}(0) = y_{0LL}$ and $y_{UU}(0) = y_{0UU}$ in Equations (20) and (21), we obtain $c_3 + c_4 = y_{0LL} - \frac{M_{UU}}{k_{LL}}$ and $c_3 - c_4 = \frac{M_{LL} - k_{UU}y_{0UU}}{\sqrt{k_{LL}k_{UU}}} = -\left(y_{0UU} - \frac{M_{LL}}{k_{UU}}\right)\sqrt{\frac{k_{UU}}{k_{LL}}}$. Therefore, the values of the integrating constants are obtained as follows:

$$c_{3} = \frac{1}{2} \left[\left(y_{0LL} - \frac{M_{UU}}{k_{LL}} \right) - \left(y_{0UU} - \frac{M_{LL}}{k_{UU}} \right) \sqrt{\frac{k_{UU}}{k_{LL}}} \right]$$
$$c_{4} = \frac{1}{2} \left[\left(y_{0LL} - \frac{M_{UU}}{k_{LL}} \right) + \left(y_{0UU} - \frac{M_{LL}}{k_{UU}} \right) \sqrt{\frac{k_{UU}}{k_{LL}}} \right]$$

Therefore, the solutions of the crisp differential Equations (16) and (19) can be obtained as follows:

$$y_{LL}(x) = \left(y_{0LL} - \frac{M_{UU}}{k_{LL}}\right) \cosh\left(\sqrt{k_{LL}k_{UU}}x\right) - \left(y_{0UU} - \frac{M_{LL}}{k_{UU}}\right) \sqrt{\frac{k_{UU}}{k_{LL}}} \sinh\left(\sqrt{k_{LL}k_{UU}}x\right) + \frac{M_{UU}}{k_{LL}}$$
$$y_{UU}(x) = \left(y_{0UU} - \frac{M_{LL}}{k_{UU}}\right) \cosh\left(\sqrt{k_{LL}k_{UU}}x\right) - \left(y_{0LL} - \frac{M_{UU}}{k_{LL}}\right) \sqrt{\frac{k_{LL}}{k_{UU}}} \sinh\left(\sqrt{k_{LL}k_{UU}}x\right) + \frac{M_{UU}}{k_{UU}}$$

Similarly, by solving the crisp differential Equations (17) and (18), we obtain the following results:

$$y_{LU}(x) = \left(y_{0LU} - \frac{M_{UL}}{k_{LU}}\right) \cosh\left(\sqrt{k_{LU}k_{UL}}x\right) - \left(y_{0UL} - \frac{M_{LU}}{k_{UL}}\right) \sqrt{\frac{k_{UL}}{k_{LU}}} \sinh\left(\sqrt{k_{LU}k_{UL}}x\right) + \frac{M_{UL}}{k_{LU}}$$
$$y_{UL}(x) = \left(y_{0UL} - \frac{M_{LU}}{k_{UL}}\right) \cosh\left(\sqrt{k_{LU}k_{UL}}x\right) - \left(y_{0LU} - \frac{M_{UL}}{k_{LU}}\right) \sqrt{\frac{k_{LU}}{k_{UL}}} \sinh\left(\sqrt{k_{LU}k_{UL}}x\right) + \frac{M_{LU}}{k_{UL}}$$

Subcase 1.4. When both k_2 and M_2 are negative type-2 interval numbers

In this scenario, the type-2 interval-valued differential equation in System (13) can be replaced by the equation incorporating four components as follows:

$$\left[\left[\frac{dy_{LL}}{dx}, \frac{dy_{LU}}{dx}\right], \left[\frac{dy_{UL}}{dx}, \frac{dy_{UU}}{dx}\right]\right] = \left[\left[-k_{UU}y_{UU} - M_{UU}, -k_{UL}y_{UL} - M_{UL}\right], \left[-k_{LU}y_{LU} - M_{LU}, -k_{LL}y_{LL} - M_{LL}\right]\right]$$

This gives four differential equations, which are given below,

$$\frac{dy_{LL}}{dx} = -k_{UU}y_{UU} - M_{UU} \tag{22}$$

$$\frac{dy_{LU}}{dx} = -k_{UL}y_{UL} - M_{UL} \tag{23}$$

$$\frac{dy_{UL}}{dx} = -k_{LU}y_{LU} - M_{LU} \tag{24}$$

$$\frac{dy_{UU}}{dx} = -k_{LL}y_{LL} - M_{LL} \tag{25}$$

Following the process detailed in the preceding Subcase 1.3, the four components of the solution are obtained as follows:

$$y_{LL}(x) = \left(y_{0LL} + \frac{M_{LL}}{k_{LL}}\right)\cosh\left(\sqrt{k_{LL}k_{UU}}x\right) - \left(y_{0UU} + \frac{M_{UU}}{k_{UU}}\right)\sqrt{\frac{k_{UU}}{k_{LL}}}\sinh\left(\sqrt{k_{LL}k_{UU}}x\right) - \frac{M_{LL}}{k_{LL}}$$
$$y_{LU}(x) = \left(y_{0LU} + \frac{M_{LU}}{k_{UU}}\right)\cosh\left(\sqrt{k_{LU}k_{UL}}x\right) - \left(y_{0UL} + \frac{M_{UL}}{k_{UL}}\right)\sqrt{\frac{k_{UU}}{k_{LU}}}\sinh\left(\sqrt{k_{LU}k_{UL}}x\right) - \frac{M_{LU}}{k_{UU}}$$
$$y_{UL}(x) = \left(y_{0UL} + \frac{M_{UL}}{k_{UL}}\right)\cosh\left(\sqrt{k_{LU}k_{UL}}x\right) - \left(y_{0LU} + \frac{M_{LU}}{k_{LU}}\right)\sqrt{\frac{k_{LU}}{k_{UL}}}\sinh\left(\sqrt{k_{LU}k_{UL}}x\right) - \frac{M_{UL}}{k_{UL}}$$
$$y_{UU}(x) = \left(y_{0UU} + \frac{M_{UU}}{k_{UU}}\right)\cosh\left(\sqrt{k_{LL}k_{UU}}x\right) - \left(y_{0LL} + \frac{M_{LL}}{k_{LL}}\right)\sqrt{\frac{k_{LL}}{k_{UU}}}\sinh\left(\sqrt{k_{LL}k_{UU}}x\right) - \frac{M_{UU}}{k_{UU}}$$

Case 2. When $\widetilde{y_2}$ is generalized Hukuhara differentiable of the second type

In this case, we consider four subcases concerning the signs of the type-2 interval numbers $\tilde{k_2}$ and $\tilde{M_2}$.

Subcase 2.1. When both k_2 and M_2 are positive type-2 interval numbers

In this scenario, the type-2 interval-valued differential equation in System (13) can be replaced by the equation incorporating four components as follows:

$$\left[\left[\frac{dy_{UU}}{dx}, \frac{dy_{UL}}{dx}\right], \left[\frac{dy_{LU}}{dx}, \frac{dy_{LL}}{dx}\right]\right] = \left[\left[k_{LL}y_{LL} + M_{LL}, k_{LU}y_{LU} + M_{LU}\right], \left[k_{UL}y_{UL} + M_{UL}, k_{UU}y_{UU} + M_{UU}\right]\right]$$

This above equation gives four differential equations, which are given below.

$$\frac{dy_{UU}}{dx} = k_{LL}y_{LL} + M_{LL} \tag{26}$$

$$\frac{dy_{UL}}{dx} = k_{LU}y_{LU} + M_{LU} \tag{27}$$

$$\frac{dy_{LU}}{dx} = k_{UL}y_{UL} + M_{UL} \tag{28}$$

$$\frac{dy_{LL}}{dx} = k_{UU}y_{UU} + M_{UU} \tag{29}$$

From Equations (26) and (29) of the above system, we obtain

$$\frac{d^2 y_{LL}}{dx^2} = k_{UU} \frac{dy_{UU}}{dx}$$
$$\frac{d^2 y_{LL}}{dx^2} = k_{UU} [k_{LL} y_{LL} + M_{LL}]$$
$$\frac{d^2 y_{LL}}{dx^2} - k_{LL} k_{UU} y_{LL} = k_{UU} M_{LL}$$

.

Therefore, the solution is

$$y_{LL}(x) = c_7 e^{\sqrt{k_{LL}k_{UU}x}} + c_8 e^{-\sqrt{k_{LL}k_{UU}x}} + \frac{1}{D^2 - k_{LL}k_{UU}} (k_{UU}M_{LL})$$
$$y_{LL}(x) = c_7 e^{\sqrt{k_{LL}k_{UU}x}} + c_8 e^{-\sqrt{k_{LL}k_{UU}x}} - \frac{M_{LL}}{k_{LL}}$$
(30)

Using Equation (30) in Equation (29), we find the value of $y_{UU}(x)$ as follows:

$$y_{UU}(x) = \frac{1}{k_{UU}} \left[-M_{UU} + \sqrt{k_{LL}k_{UU}} \left(c_7 e^{\sqrt{k_{LL}k_{UU}}x} - c_8 e^{-\sqrt{k_{LL}k_{UU}}x} \right) \right]$$
(31)

Using the initial conditions $y_{LL}(0) = y_{0LL}$ and $y_{UU}(0) = y_{0UU}$ in Equations (32) and (31), we obtain $c_7 + c_8 = y_{0LL} + \frac{M_{LL}}{k_{LL}}$ and $c_7 - c_8 = \left(y_{0UU} + \frac{M_{UU}}{k_{UU}}\right) \sqrt{\frac{k_{UU}}{k_{LL}}}$. Therefore, the values of the integrating constants are obtained as follows:

$$c_{7} = \frac{1}{2} \left[\left(y_{0LL} + \frac{M_{LL}}{k_{LL}} \right) + \left(y_{0UU} + \frac{M_{UU}}{k_{UU}} \right) \sqrt{\frac{k_{UU}}{k_{LL}}} \right]$$
$$c_{8} = \frac{1}{2} \left[\left(y_{0LL} + \frac{M_{LL}}{k_{LL}} \right) - \left(y_{0UU} + \frac{M_{UU}}{k_{UU}} \right) \sqrt{\frac{k_{UU}}{k_{LL}}} \right]$$

Therefore, the solutions of the crisp differential Equations (26) and (29) can be obtained as follows:

$$y_{LL}(x) = \left(y_{0LL} + \frac{M_{LL}}{k_{LL}}\right) \cosh\left(\sqrt{k_{LL}k_{UU}}x\right) + \left(y_{0UU} + \frac{M_{UU}}{k_{UU}}\right) \sqrt{\frac{k_{UU}}{k_{LL}}} \sinh\left(\sqrt{k_{LL}k_{UU}}x\right) - \frac{M_{LL}}{k_{LL}}$$
$$y_{UU}(x) = \left(y_{0UU} + \frac{M_{UU}}{k_{UU}}\right) \cosh\left(\sqrt{k_{LL}k_{UU}}x\right) + \left(y_{0LL} + \frac{M_{LL}}{k_{LL}}\right) \sqrt{\frac{k_{LL}}{k_{UU}}} \sinh\left(\sqrt{k_{LL}k_{UU}}x\right) - \frac{M_{UU}}{k_{UU}}$$

Similarly, by solving the crisp differential Equations (27) and (28), we obtain the following results:

$$y_{LU}(x) = \left(y_{0LU} + \frac{M_{LU}}{k_{LU}}\right) \cosh\left(\sqrt{k_{LU}k_{UL}}x\right) + \left(y_{0UL} + \frac{M_{UL}}{k_{UL}}\right) \sqrt{\frac{k_{UL}}{k_{LU}}} \sinh\left(\sqrt{k_{LU}k_{UL}}x\right) - \frac{M_{LU}}{k_{LU}}$$
$$y_{UL}(x) = \left(y_{0UL} + \frac{M_{UL}}{k_{UL}}\right) \cosh\left(\sqrt{k_{LU}k_{UL}}x\right) + \left(y_{0LU} + \frac{M_{LU}}{k_{LU}}\right) \sqrt{\frac{k_{LU}}{k_{UL}}} \sinh\left(\sqrt{k_{LU}k_{UL}}x\right) - \frac{M_{UL}}{k_{UL}}$$

Subcase 2.2. When k_2 and M_2 are positive and negative type-2 interval numbers, respectively In this scenario, the type-2 interval-valued differential equation in System (13) can be replaced by the equation incorporating four components as follows:

$$\left[\left[\frac{dy_{UU}}{dx}, \frac{dy_{UL}}{dx}\right], \left[\frac{dy_{LU}}{dx}, \frac{dy_{LL}}{dx}\right]\right] = \left[\left[k_{LL}y_{LL} - M_{UU}, k_{LU}y_{LU} - M_{UL}\right], \left[k_{UL}y_{UL} - M_{LU}, k_{UU}y_{UU} - M_{LL}\right]\right]$$

This above equation gives four differential equations, which are given below.

$$\frac{dy_{UU}}{dx} = k_{LL}y_{LL} - M_{UU} \tag{32}$$

$$\frac{dy_{UL}}{dx} = k_{LU}y_{LU} - M_{UL} \tag{33}$$

$$\frac{dy_{LU}}{dx} = k_{UL}y_{UL} - M_{LU} \tag{34}$$

$$\frac{dy_{LL}}{dx} = k_{UU}y_{UU} - M_{LL} \tag{35}$$

Following the process detailed in the preceding Subcase 2.1, the four components of the solution are obtained as follows:

$$y_{LL}(x) = \left(y_{0LL} - \frac{M_{UU}}{k_{LL}}\right) \cosh\left(\sqrt{k_{LL}k_{UU}}x\right) + \left(y_{0UU} - \frac{M_{LL}}{k_{UU}}\right)\sqrt{\frac{k_{UU}}{k_{LL}}}\sinh\left(\sqrt{k_{LL}k_{UU}}x\right) + \frac{M_{UU}}{k_{LL}}$$
$$y_{LU}(x) = \left(y_{0LU} - \frac{M_{UL}}{k_{LU}}\right)\cosh\left(\sqrt{k_{LU}k_{UL}}x\right) + \left(y_{0UL} - \frac{M_{LU}}{k_{UL}}\right)\sqrt{\frac{k_{UL}}{k_{LU}}}\sinh\left(\sqrt{k_{LU}k_{UL}}x\right) + \frac{M_{UL}}{k_{LU}}$$
$$y_{UL}(x) = \left(y_{0UL} - \frac{M_{LU}}{k_{UL}}\right)\cosh\left(\sqrt{k_{LU}k_{UL}}x\right) + \left(y_{0LU} - \frac{M_{UL}}{k_{LU}}\right)\sqrt{\frac{k_{UL}}{k_{LU}}}\sinh\left(\sqrt{k_{LU}k_{UL}}x\right) + \frac{M_{LU}}{k_{UL}}$$
$$y_{UU}(x) = \left(y_{0UU} - \frac{M_{LL}}{k_{UU}}\right)\cosh\left(\sqrt{k_{LL}k_{UU}}x\right) + \left(y_{0LL} - \frac{M_{UU}}{k_{LL}}\right)\sqrt{\frac{k_{UL}}{k_{UU}}}\sinh\left(\sqrt{k_{LL}k_{UU}}x\right) + \frac{M_{LL}}{k_{UU}}$$

Subcase 2.3. When k_2 and M_2 are negative and positive type-2 interval numbers, respectively In this scenario, the type-2 interval-valued differential equation in System (13) can be replaced by the equation incorporating four components as follows:

$$\left[\left[\frac{dy_{UU}}{dx}, \frac{dy_{UL}}{dx}\right], \left[\frac{dy_{LU}}{dx}, \frac{dy_{LL}}{dx}\right]\right] = \left[\left[-k_{UU}y_{UU} + M_{LL}, -k_{UL}y_{UL} + M_{LU}\right], \left[-k_{LU}y_{LU} + M_{UL}, -k_{LL}y_{LL} + M_{UU}\right]\right]$$

Following the process detailed in the preceding Subcases 1.1 and 1.2, the four components of the solution are obtained as follows:

$$y_{LL} = \frac{M_{UU}}{k_{LL}} \left(1 - e^{-k_{LL}x} \right) + y_{0LL}e^{-k_{LL}x}$$
$$y_{LU} = \frac{M_{UL}}{k_{LU}} \left(1 - e^{-k_{LU}x} \right) + y_{0LU}e^{-k_{LU}x}$$
$$y_{UL} = \frac{M_{LU}}{k_{UL}} \left(1 - e^{-k_{UL}x} \right) + y_{0UL}e^{-k_{UL}x}$$
$$y_{UU} = \frac{M_{LL}}{k_{UU}} \left(1 - e^{-k_{UU}x} \right) + y_{0UU}e^{-k_{UU}x}$$

Subcase 2.4. When both k_2 and M_2 are negative type-2 interval numbers

In this scenario, the type-2 interval-valued differential equation in System (13) can be replaced by the equation incorporating four components as follows:

$$\left[\left[\frac{dy_{UU}}{dx}, \frac{dy_{UL}}{dx}\right], \left[\frac{dy_{LU}}{dx}, \frac{dy_{LL}}{dx}\right]\right] = \left[\left[-k_{UU}y_{UU} - M_{UU}, -k_{UL}y_{UL} - M_{UL}\right], \left[-k_{LU}y_{LU} - M_{LU}, -k_{LL}y_{LL} - M_{LL}\right]\right]$$

Following the process detailed in the preceding Subcases 1.1 and 1.2, the four components of the solution are obtained as follows:

$$y_{LL}(x) = \frac{M_{LL}}{k_{LL}} \left(e^{-k_{LL}x} - 1 \right) + y_{0LL} e^{-k_{LL}x}$$
$$y_{LU}(x) = \frac{M_{LU}}{k_{LU}} \left(e^{-k_{LU}x} - 1 \right) + y_{0LU} e^{-k_{LU}x}$$
$$y_{UL}(x) = \frac{M_{UL}}{k_{UL}} \left(e^{-k_{UL}x} - 1 \right) + y_{0UL} e^{-k_{UL}x}$$
$$y_{UU}(x) = \frac{M_{UU}}{k_{UU}} \left(e^{-k_{UU}x} - 1 \right) + y_{0UU} e^{-k_{UU}x}$$

5. Inventory Control Problem as an Application

Several scholars have looked at the application of the interval number theory to inventory managerial challenges. Interval numbers express the disparity of ambiguity in a range of values. In certain instances of mathematical modeling, the upper and lower ends of the interval numbers may be flexible. For example, the demand pattern fluctuates between ranges in a storefront setting. Sometimes, the ranges' limitations are also variable. To address this scenario, we must broaden the definition of uncertainty as provided by interval numbers. Therefore, this section discusses an economic order quantity model under type-2 interval uncertainty. The hypotheses of the model are as follows:

- 1. Demand is a linear function of price and stock, i.e., as time goes on, the demand rate increases linearly. $D(t) = a b\tilde{p}_2 + c\tilde{I}_2(t)$, where \tilde{p}_2 is the selling price per unit, $\tilde{I}_2(t)$ is the stock as type-2 interval numbers, and *a*, *b*, *c* are positive crisp constants.
- 2. The deterioration rate θ_2 is constant and assumed to be a type-2 interval number.
- 3. No shortage is allowed.
- 4. The replenishment rate is infinite, but the lot size is finite.
- 5. The time horizon is finite.
- 6. The lead time is zero.

5.1. Model Formulation and Discussion

Let us describe an EOQ model and assume that the initial stock level Q_2 is the starting point for the proposed inventory model with deteriorated items. Due to the combined effects of demand and deterioration, a progressive stock level declined over the entire period. Following the gradual decay, the inventory level drops to zero at the end of the lot cycle, i.e., at the time t = T. The following type-2 interval-valued differential equation is the mathematical representative of the proposed EOQ model:

$$\frac{dI_2}{dt} + \overset{\sim}{\theta_2} \overset{\sim}{I_2}(t) = -\left\{a - b \overset{\sim}{p_2} + c \overset{\sim}{I_2}(t)\right\}$$
(36)

with $\widetilde{I}_2(0) = \widetilde{Q}_2$ and $I_2(T) = 0$.

Case 1. When $I_2(t)$ is generalized Hukuhara differentiable of the first type

In this case, Equation (36) can be written as follows:

$$\begin{bmatrix} \begin{bmatrix} dI_{LL} \\ dt \end{bmatrix}, \begin{bmatrix} dI_{UL} \\ dt \end{bmatrix}, \begin{bmatrix} dI_{UL} \\ dt \end{bmatrix}, \begin{bmatrix} dI_{UL} \\ dt \end{bmatrix} \end{bmatrix} + [[\theta_{LL}, \theta_{LU}], [\theta_{UL}, \theta_{UU}]][[I_{LL}, I_{LU}], [I_{UL}, I_{UU}]] \\ = -a + b[[p_{LL}, p_{LU}], [p_{UL}, p_{UU}]] - c[[I_{LL}, I_{LU}], [I_{UL}, I_{UU}]]$$

This preceding expression gives four differential equations, which are given below.

$$\frac{dI_{LL}}{dt} + \theta_{LL}I_{LL} = -a + bp_{LL} - cI_{UU}$$
(37)

$$\frac{dI_{LU}}{dt} + \theta_{LU}I_{LU} = -a + bp_{LU} - cI_{UL}$$
(38)

$$\frac{dI_{UL}}{dt} + \theta_{UL}I_{UL} = -a + bp_{UL} - cI_{LU}$$
(39)

$$\frac{dI_{UU}}{dt} + \theta_{UU}I_{UU} = -a + bp_{UU} - cI_{LL}$$
(40)

From Equations (37) and (40), we obtain a system of crisp differential equation as given below.

$$\begin{cases}
\frac{dI_{LL}}{dt} = -\theta_{LL}I_{LL} - cI_{UU} + \alpha_1 \\
\frac{dI_{UU}}{dt} = -cI_{LL} - \theta_{UU}I_{UU} + \alpha_4 \\
I_{LL}(T) = 0, I_{UU}(T) = 0
\end{cases}$$
(41)

where $\alpha_1 = -a + bp_{LL}$, $\alpha_4 = -a + bp_{UU}$ and from Equations (38) and (39), we obtain a system of crisp differential equation as given below.

$$\begin{cases} \frac{dI_{LU}}{dt} = -\theta_{LU}I_{LU} - cI_{UL} + \alpha_2\\ \frac{dI_{UL}}{dt} = -CI_{LU} - \theta_{UL}I_{UL} + \alpha_3\\ I_{LU}(T) = 0, I_{UL}(T) = 0 \end{cases}$$
(42)

where $\alpha_2 = -a + bp_{LU}$ and $\alpha_3 = -a + bp_{UL}$. We solve System (41) using Lagrange's multiplier method in the following way:

4

 $d(I_{LL} + \lambda I_{IIII})$

$$\frac{d(I_{LL} + \lambda I_{UU})}{dt} = -(\theta_{LL} + \lambda c)I_{LL} - (c + \lambda\theta_{UU})I_{UU} + \alpha_1 + \lambda\alpha_4$$

i.e.,
$$\frac{d(I_{LL} + \lambda I_{UU})}{dt} = -(\theta_{LL} + \lambda c)\left\{I_{LL} + \frac{(c + \lambda\theta_{UU})}{(\theta_{LL} + \lambda c)}I_{UU} - \frac{\alpha_1 + \lambda\alpha_4}{(\theta_{LL} + \lambda c)}\right\}$$
(43)

We chose a λ such that $\frac{(c+\lambda\theta_{UU})}{(\theta_{LL}+\lambda c)} = \lambda$, which gives two different values of λ , say λ_1 and λ_2 , and Equation (43) becomes

$$\frac{d(I_{LL} + \lambda I_{UU})}{dt} = -(\theta_{LL} + \lambda c) \left\{ I_{LL} + \lambda I_{UU} - \frac{\alpha_1 + \lambda \alpha_4}{(\theta_{LL} + \lambda c)} \right\}$$

i.e., $\frac{du(t)}{dt} = -(\theta_{LL} + \lambda c)u(t)$, where $u(t) = I_{LL} + \lambda I_{UU} - \frac{\alpha_1 + \lambda \alpha_2}{(\theta_{LL} + \lambda c)}$

or $u(t) = Ae^{-(\theta_{LL} + \lambda c)t}$

$$I_{LL} + \lambda I_{UU} - \frac{\alpha_1 + \lambda \alpha_4}{(\theta_{LL} + \lambda c)} = A e^{-(\theta_{LL} + \lambda c)t}$$

This satisfies the initial conditions $I_{LL}(T) = 0$ and $I_{UU}(T) = 0$.

$$I_{LL} + \lambda I_{UU} = \frac{\alpha_1 + \lambda \alpha_4}{(\theta_{LL} + \lambda c)} \left(1 - e^{(\theta_{LL} + \lambda c)(T-t)} \right)$$

For two values of λ_1 and λ_2 , we obtain two simultaneous equations:

$$I_{LL} + \lambda_1 I_{UU} = M_1 \left(1 - e^{N_1 (T - t)} \right)$$
(44)

$$I_{LL} + \lambda_2 I_{UU} = M_2 \left(1 - e^{N_2 (T-t)} \right)$$
(45)

where $M_1 = \frac{\alpha_1 + \lambda_1 \alpha_4}{(\theta_{LL} + \lambda_1 c)}$, $N_1 = (\theta_{LL} + \lambda_1 c)$, $M_2 = \frac{\alpha_1 + \lambda_2 \alpha_4}{(\theta_{LL} + \lambda_2 c)}$, and $N_2 = (\theta_{LL} + \lambda_2 c)$. Solving Equations (44) and (45), we obtain the solution of System (41).

$$\begin{cases} I_{LL}(t) = \frac{M_1 \lambda_2 (1 - e^{N_1 (T-t)}) - M_2 \lambda_1 (1 - e^{N_2 (T-t)})}{\lambda_2 - \lambda_1} \\ I_{UU}(t) = \frac{M_1 (1 - e^{N_1 (T-t)}) - M_2 (1 - e^{N_2 (T-t)})}{\lambda_1 - \lambda_2} \\ 0 \le t \le T \end{cases}$$
(46)

Using the initial conditions $I_{LL}(0) = Q_{LL}$ and $I_{UU}(0) = Q_{UU}$ in Equation (46), we obtain two components of the type-2 interval-valued order size as given below.

$$Q_{LL} = \frac{M_1 \lambda_2 (1 - e^{N_1 T}) - M_2 \lambda_1 (1 - e^{N_2 T})}{\lambda_2 - \lambda_1}$$
$$Q_{UU} = \frac{M_1 (1 - e^{N_1 T}) - M_2 (1 - e^{N_2 T})}{\lambda_1 - \lambda_2}$$

Similarly, solving System (42) by Lagrange's multiplier method

$$I_{LU} + \mu I_{UL} = \frac{\alpha_2 + \mu \alpha_3}{(\theta_{LU} + \mu c)} \left(1 - e^{(\theta_{LU} + \mu c)(T-t)} \right) \text{ where } \mu = \frac{(c + \mu \theta_{UL})}{(\theta_{LU} + \mu c)}$$

So, for μ_1 and μ_2 , we obtained the solutions as

$$\begin{cases} I_{LU}(t) = \frac{M_3 \mu_2 (1 - e^{N_3 (T-t)}) - M_4 \mu_1 (1 - e^{N_4 (T-t)})}{\mu_2 - \mu_1} \\ I_{UL}(t) = \frac{M_3 (1 - e^{N_3 (T-t)}) - M_4 (1 - e^{N_4 (T-t)})}{\mu_1 - \mu_2} \\ 0 \le t \le T \end{cases}$$
(47)

where
$$M_3 = \frac{\alpha_2 + \mu_1 \alpha_3}{(\theta_{LU} + \mu_1 c)}$$
, $N_3 = (\theta_{LU} + \mu_1 c)$, $M_4 = \frac{\alpha_2 + \mu_2 \alpha_3}{(\theta_{LU} + \mu_2 c)}$, and $N_4 = (\theta_{LU} + \mu_2 c)$.

Using the initial conditions $I_{LU}(0) = Q_{LU}$ and $I_{UL}(0) = Q_{UL}$ in System (47), we obtain two components of the type-2 interval-valued order size as given below.

$$Q_{LU} = \frac{M_3\mu_2(1-e^{N_3T}) - M_4\mu_1(1-e^{N_4T})}{\mu_2 - \mu_1}$$
$$Q_{UL} = \frac{M_3(1-e^{N_3T}) - M_4(1-e^{N_4T})}{\mu_1 - \mu_2}$$

Several relevant costs and the earned revenue will be obtained as follows:

- (i) The replenishment cost is constant and is taken to be *K*.
- (ii) Holding cost: Let $hc = [[hc_{LL}, hc_{LU}], [hc_{UL}, hc_{UU}]]$ be the per unit holding cost per unit of time. Then, the holding cost $HC_2 = [[HC_{LL}, HC_{LU}], [HC_{UL}, HC_{UU}]]$ is given by

$$\begin{split} HC_{LL} &= hc_{LL} \int_{0}^{T} I_{LL}(t) dt = hc_{LL} \int_{0}^{T} \frac{M_1 \lambda_2 \left(1 - e^{N_1(T-t)}\right) - M_2 \lambda_1 \left(1 - e^{N_2(T-t)}\right)}{\lambda_2 - \lambda_1} dt \\ &= \frac{hc_{LL}}{\lambda_2 - \lambda_1} \left[\frac{M_1 \lambda_2}{N_1} \left(N_1 T + 1 - e^{N_1 T}\right) - \frac{M_2 \lambda_1}{N_2} \left(N_2 T + 1 - e^{N_2 T}\right) \right] \end{split}$$

$$\begin{aligned} HC_{LU} &= hc_{LU} \int_{0}^{T} I_{LU}(t) dt = hc_{LU} \int_{0}^{T} \frac{M_{3}\mu_{2} \left(1 - e^{N_{3}(T-t)}\right) - M_{4}\mu_{1} \left(1 - e^{N_{4}(T-t)}\right)}{\mu_{2} - \mu_{1}} dt \\ &= \frac{hc_{LU}}{\mu_{2} - \mu_{1}} \left[\frac{M_{3}\mu_{2}}{N_{3}} \left(N_{3}T + 1 - e^{N_{3}T}\right) - \frac{M_{4}\mu_{1}}{N_{4}} \left(N_{4}T + 1 - e^{N_{4}T}\right) \right] \end{aligned}$$

$$\begin{aligned} HC_{UL} &= hc_{UL} \int_{0}^{T} I_{UL}(t) dt = hc_{UL} \int_{0}^{T} \frac{M_3 \left(1 - e^{N_3 (T-t)}\right) - M_4 \left(1 - e^{N_4 (T-t)}\right)}{\mu_1 - \mu_2} dt \\ &= \frac{hc_{UL}}{\mu_1 - \mu_2} \left[\frac{M_3}{N_3} \left(N_3 T + 1 - e^{N_3 T}\right) - \frac{M_4}{N_4} \left(N_4 T + 1 - e^{N_4 T}\right) \right] \end{aligned}$$

$$\begin{aligned} HC_{UU} &= hc_{UU} \int_{0}^{T} I_{UU}(t) dt = hc_{UU} \int_{0}^{T} \frac{M_1 \left(1 - e^{N_1(T-t)}\right) - M_2 \left(1 - e^{N_2(T-t)}\right)}{\lambda_1 - \lambda_2} dt \\ &= \frac{hc_{UU}}{\lambda_1 - \lambda_2} \left[\frac{M_1}{N_1} \left(N_1 T + 1 - e^{N_1 T}\right) - \frac{M_2}{N_2} \left(N_2 T + 1 - e^{N_2 T}\right) \right] \end{aligned}$$

(iii) Purchase cost: Let $m = [[m_{LL}, m_{LU}], [m_{UL}, m_{UU}]]$ be the per unit purchasing cost $\stackrel{\sim}{PC_2} = [[PC_{LL}, PC_{LU}], [PC_{UL}, PC_{UU}]]$ is given by

$$PC_{LL} = m_{LL}Q_{LL} = m_{LL}\frac{M_1\lambda_2(1-e^{N_1T}) - M_2\lambda_1(1-e^{N_2T})}{\lambda_2 - \lambda_1}$$
$$PC_{LU} = m_{LU}Q_{LU} = m_{LU}\frac{M_3\mu_2(1-e^{N_3T}) - M_4\mu_1(1-e^{N_4T})}{\mu_2 - \mu_1}$$

$$PC_{UL} = m_{UL}Q_{UL} = m_{UL}\frac{M_3(1 - e^{N_3T}) - M_4(1 - e^{N_4T})}{\mu_1 - \mu_2}$$
$$PC_{UU} = m_{UU}Q_{UU} = m_{UU}\frac{M_1(1 - e^{N_1T}) - M_2(1 - e^{N_2T})}{\lambda_1 - \lambda_2}$$

(iv) The total sales revenue is $\widetilde{SR}_2 = [[SR_{LL}, SR_{LU}], [SR_{UL}, SR_{UU}]]$ during the entire cycle. Then,

$$SR_{LL} = p_{LL} \int_{0}^{T} \{a - bp_{UU} + cI_{LL}(t)\}dt$$

$$= p_{LL}(a - bp_{UU})T + \frac{p_{LL}c}{\lambda_2 - \lambda_1} \left[\frac{M_1\lambda_2}{N_1} \left(N_1T + 1 - e^{N_1T} \right) - \frac{M_2\lambda_1}{N_2} \left(N_2T + 1 - e^{N_2T} \right) \right]$$
$$SR_{LU} = p_{LU} \int_0^T \{a - bp_{UL} + cI_{LU}(t)\} dt$$

$$= p_{LU}(a - bp_{UL})T + \frac{p_{LU}c}{\mu_2 - \mu_1} \left[\frac{M_3\mu_2}{N_3} \left(N_3T + 1 - e^{N_3T} \right) - \frac{M_4\mu_1}{N_4} \left(N_4T + 1 - e^{N_4T} \right) \right]$$
$$SR_{UL} = p_{UL} \int_0^T \{a - bp_{LU} + cI_{UL}(t)\} dt$$

$$= p_{UL}(a - b_{LU})T + \frac{p_{UL}c}{\mu_1 - \mu_2} \left[\frac{M_3}{N_3} \left(N_3T + 1 - e^{N_3T} \right) - \frac{M_4}{N_4} \left(N_4T + 1 - e^{N_4T} \right) \right]$$
$$SR_{UU} = p_{UU} \int_0^T \{ a - bp_{LL} + cI_{UU}(t) \} dt$$

$$= p_{UU}(a - bp_{LL})T + \frac{p_{UU}c}{\lambda_1 - \lambda_2} \left[\frac{M_1}{N_1} \left(N_1T + 1 - e^{N_1T} \right) - \frac{M_2}{N_2} \left(N_2T + 1 - e^{N_2T} \right) \right]$$

Therefore, the total average profit $\widetilde{TP}_2 = [[TP_{1LL}, TP_{1LU}], [TP_{1UL}, TP_{1UU}]]$ of the system during the entire cycle is given by: $TP_{1LL} = \frac{SR_{LL} - K - HC_{UU} - PC_{UU}}{T}, TP_{1LU} = \frac{SR_{LU} - K - HC_{UL} - PC_{UL}}{T}, TP_{1UL} = \frac{SR_{UL} - K - HC_{LU} - PC_{LU}}{T},$ and $TP_{1UU} = \frac{SR_{UU} - K - HC_{LL} - PC_{LL}}{T}.$

Therefore, the optimization problem for the proposed model can be written mathematically in the following form:

. .

$$\begin{cases}
Max TP_{1LL} \\
Max TP_{1LU} \\
Max TP_{1UL} \\
Max TP_{1UU} \\
0 \le t \le T
\end{cases}$$
(48)

Case 2. When $\widetilde{I_2}(t)$ is generalized Hukuhara differentiable of the second type

In such cases,

$$\begin{bmatrix} \begin{bmatrix} dI_{UU} \\ dt \end{bmatrix}, \begin{bmatrix} dI_{LU} \\ dt \end{bmatrix}, \begin{bmatrix} dI_{LU} \\ dt \end{bmatrix} \end{bmatrix} + [[\theta_{LL}, \theta_{LU}], [\theta_{UL}, \theta_{UU}]][[I_{LL}, I_{LU}], [I_{UL}, I_{UU}]] \\ = -a + b[[p_{LL}, p_{LU}], [p_{UL}, p_{UU}]] - c[[I_{LL}, I_{LU}], [I_{UL}, I_{UU}]]$$

This gives four differential equations, which are given below.

$$\frac{dI_{UU}}{dt} + \theta_{LL}I_{LL} = -a + bp_{LL} - cI_{UU}$$
(49)

$$\frac{dI_{UL}}{dt} + \theta_{LU}I_{LU} = -a + bp_{LU} - cI_{UL}$$
(50)

$$\frac{dI_{LU}}{dt} + \theta_{UL}I_{UL} = -a + bp_{UL} - cI_{LU}$$
(51)

$$\frac{dI_{LL}}{dt} + \theta_{UU}I_{UU} = -a + bp_{UU} - cI_{LL}$$
(52)

From Equations (49) and (52), we obtain the following system of crisp differential equation with initial conditions,

$$\begin{cases} \frac{dI_{LL}}{dt} = -cI_{LL} - \theta_{UU}I_{UU} + \alpha_4 \\ \frac{dI_{UU}}{dt} = -\theta_{LL}I_{LL} - cI_{UU} + \alpha_1 \\ I_{LL}(0) = Q_{LL}, I_{UU}(0) = Q_{UU} \end{cases}$$
(53)

and from Equations (50) and (51), we obtain the following system of crisp differential equation with initial conditions,

$$\begin{cases} \frac{dI_{LU}}{dt} = -cI_{LU} - \theta_{UL}I_{UL} + \alpha_3\\ \frac{dI_{UL}}{dt} = -\theta_{LU}I_{LU} - cI_{UL} + \alpha_2\\ I_{LU}(0) = Q_{LU}, I_{UL}(0) = Q_{UL} \end{cases}$$
(54)

By solving Systems (53) and (54) by applying the process as in Case 1, we obtain the total average profit $\widetilde{TP}_2 = [[TP_{2LL}, TP_{2LU}], [TP_{2UL}, TP_{2UU}]]$ of the system during the entire cycle, which is given by: $TP_{2LL} = \frac{SR_{LL}-K-HC_{UU}-PC_{UU}}{T}, TP_{2LU} = \frac{SR_{LU}-K-HC_{UL}-PC_{UL}}{T}, TP_{2UL} = \frac{SR_{UL}-K-HC_{LU}-PC_{LU}}{T}$ and $TP_{2UU} = \frac{SR_{UU}-K-HC_{LL}-PC_{LL}}{T}$. (See Appendix A).

Therefore, the optimization problem for the proposed model can be written mathematically in the following form:

$$Max TP_{2LL}$$

$$Max TP_{2LU}$$

$$Max TP_{2UL}$$

$$Max TP_{2UU}$$

$$0 \le t \le T$$
(55)

5.2. Numerical Results and Its Graphical Display

The following data are considered as inputs for the numerical optimization of the models for both cases:

 $a = 300, b = 0.1, p_{LL} = 15, p_{LU} = 16, p_{UL} = 18, p_{UU} = 19, c = 0.15, \theta_{LL} = 0.045, \theta_{LU} = 0.05, \theta_{UL} = 0.06, \theta_{UU} = 0.065.$ $K = 300, p_{c_{LL}} = 4.5, p_{c_{LU}} = 5, p_{c_{UL}} = 5.5, p_{c_{UU}} = 6, h_{LL} = 2.5, h_{LU} = 3, h_{UL} = 3.5, h_{UU} = 4.$

Then, the optimum results for the two cases are as follows:

Case 1. When $\widetilde{I_2}(t)$ is generalized Hukuhara differentiable of the first type

Then, the values of intermediate parameters as follows: $\alpha_1 = -298.5$, $\alpha_4 = -298.1$, $\alpha_2 = -298.4$, $\alpha_3 = -298.2$, $\lambda_1 = 1.06889$, $\lambda_2 = -0.935553$, $M_1 = -3005.530558822599$, $M_2 = 205.71744291978783$, $N_1 = 0.2053335$, $N_2 = -0.09533294999999999$, $\mu_1 = 1.03389$, $\mu_2 = -0.967222$, $M_3 = -2958.33647270502$, $M_4 = 104.90169777447744$, $N_3 = 0.2050834999999997$, and $N_4 = -0.0950833$. For the optimal values of the objective function, the total average profit is $TP_2 = [[1578.07, 2271.07], [3319.43, 4008.27]]$ with optimal lot size $Q_2 = [[646.986, 649.412], [654.310, 656.782]]$ and optimal lot cycle $T^* = 1.806$. Figure 2 represents the graph for the total average profit concerning the time cycle.



Figure 2. Total average profit (TP_2) versus lot cycle (T^*) for the generalized Hukuhara differentiability of the first type. The curves are represented by blue, yellow, green, and red for TP_{1LL} , TP_{1LU} , TP_{1UL} , and TP_{1UU} , respectively. The concave curves exhibit the optimal value of the total average profit.

Case 2. When $I_2(t)$ is generalized Hukuhara differentiable of the second type

Then, the values of intermediate parameters as follows: $\alpha_1 = -298.5$, $\alpha_4 = -298.1$, $\alpha_2 = -298.4$, $\alpha_3 = -298.2$, $\lambda_3 = 1.202$, $\lambda_4 = -1.202$, $C_1 = -3218.663334803273$, $C_2 = 632.8537170263784$, $D_1 = 0.20409$, $D_2 = 0.09591$, $\mu_3 = 1.095$, $\mu_4 = -1.095$, $C_3 = -3052.2490842490843$, $C_4 = 299.7165354330709$, $D_3 = 0.20475$, $D_4 = 0.09525$. For the optimal values of the objective function, the total average profit is $TP_2 = [[1740.72, 2372.01]$, [3334.59, 3970.55]] with optimal lot size $\tilde{Q}_2^{*} = [[549.922, 547.967]$, [544.104, 542.148]] and optimal lot cycle $T^* = 1.554$. Figure 3 represents the graph for the total average profit concerning the time cycle.

5.3. Applicability of the Proposed Model

Price has a ripple effect on customers' appetite for merchandise in the context of retailing. The buying and selling phenomenon in developing or underdeveloped nations almost immediately shows that lower prices lure purchasers to the goods. Price, then, has an impact on average profit as well. However, accurate pricing for an effective strategy design favoring the retailer's perspective possesses some sense of vagueness. The dilemma in optimal pricing can be adjusted by considering interval decision phenomena in which price can be considered as an interval number having upper and lower bounds of values. However, it is not necessarily true for a real-world business scenario that the values can fit within two specific bounds. Instead, it may be viewed in the type-2 interval number theory, which provides a broader sense of interval uncertainty. Also, showrooms have impacts on the demand regulation. The collection of active stock in the showroom creates additional demand. In the same logic, various associated costs and the stock level can be taken into account in the type-2 interval number setting. Since the impact of pricing on the demand and profit control decisions is very significant and uncertainties arise in designing real-

world strategies, the discussed theory and results can be applied in a showroom concerning retail dealing phenomena.



Figure 3. Total average profit (TP_2) versus lot cycle (T^*) for the case of generalized Hukuhara differentiability of the second type. The curves are represented by blue, yellow, green, and red for TP_{2LL} , TP_{2LU} , TP_{2UL} , and TP_{2UU} , respectively. The concave curves exhibit the optimal value of the total average profit.

6. Conclusions

Type-2 interval number theory broadens the theory of interval numbers, making the ends of an interval flexible within bounds. The arithmetic characteristics of type-2 interval numbers and their usage for optimizing mathematical models, particularly inventory models, have received focus. It is recommended to use the generalized Hukuhara differentiability technique for discussing differential equation-based mathematical models with type-2 interval uncertainty. Before discussing the solution approach, the solution's existence and its uniqueness, if it exists, must be verified. This paper contributes to the existence and uniqueness theory for solving type-2 interval-valued differential equations. Also, the solution approach of the type-2 interval-valued differential equations has been discussed in this paper. The utility of the proposed theory has been exercised on an EOQ model. The EOQ model of deteriorated items has been discussed for both cases of generalized Hukuhara differentiability. The main aim of this paper is to establish the theory of solvability of the type-2 interval-valued differential equations. The inventory model is not a priority in this paper. This research is a precursor to differential equation concepts and their utilization in a type-2 interval setting. The current form leaves room for additional investigation in the following areas:

- 1. The theory of integral equations for type-2 interval-valued functions may be introduced following the footsteps of this paper.
- 2. More implicit and explicit characteristics of type-2 interval-valued calculus may have emerged as an epicenter for future research.
- 3. This theory's fractional order counterpart may be a fruitful consequence of the proposed approach.
- 4. We have discussed a popular lot-size model as an application of the proposed theory. We have shortened our analysis because the inventory model was not our main concern. In the future, a robust approach to model formulation and optimization associated with raw data from the market will be the ultimate consequence of the proposed theory.

Author Contributions: Conceptualization, M.R., S.Z. and S.S.; Methodology, R.H., S.A. and F.A.; Writing, S.Z., S.S. and S.P.M.; Software and formal analysis, R.H. and F.A.; Validation, M.R., S.S. and F.A.; Resources and Supervision, S.A. and F.A.; Visualization, S.Z., R.H. and S.S.; Review and Editing, M.R., S.A. and S.P.M.; Funding acquisition, S.Z. All authors have read and agreed to the published version of the manuscript.

Funding: This research contain no external funding.

Data Availability Statement: The used data are already in this manuscript. The data sources are also clearly mentioned.

Conflicts of Interest: The authors declare no conflict of interest.

``

Appendix A

1/7

. . . .

We solve System (43) using Lagrange's multiplier method in the following way:

$$\frac{d(I_{LL} + \lambda I_{UU})}{dt} = -(c + \lambda \theta_{LL})I_{LL} - (\theta_{UU} + \lambda c)I_{UU} + \alpha_4 + \lambda \alpha_1$$

i.e.,
$$\frac{d(I_{LL} + \lambda I_{UU})}{dt} = -(c + \lambda \theta_{LL})\left\{I_{LL} + \frac{(\theta_{UU} + \lambda c)}{(c + \lambda \theta_{LL})}I_{UU} - \frac{\alpha_4 + \lambda \alpha_1}{(c + \lambda \theta_{LL})}\right\}$$
(A1)

We chose a λ such that $\frac{(\theta_{UU}+\lambda c)}{(c+\lambda\theta_{LL})} = \lambda$, which gives two different values of λ , say λ_3 and λ_4 , and Equation (A1) becomes

$$\frac{d(I_{LL} + \lambda I_{UU})}{dt} = -(c + \lambda \theta_{LL}) \left\{ I_{LL} + \lambda I_{UU} - \frac{\alpha_4 + \lambda \alpha_1}{(c + \lambda \theta_{LL})} \right\}$$

i.e., $\frac{du(t)}{dt} = -(c + \lambda \theta_{LL})u(t)$, where $u(t) = I_{LL} + \lambda I_{UU} - \frac{\alpha_4 + \lambda \alpha_1}{(c + \lambda \theta_{LL})}$
or $u(t) = Ae^{-(c + \lambda \theta_{LL})t}$

$$I_{LL} + \lambda I_{UU} - \frac{\alpha_4 + \lambda \alpha_1}{(c + \lambda \theta_{LL})} = A e^{-(c + \lambda \theta_{LL})t}$$

This satisfies the initial conditions $I_{LL}(T) = 0$ and $I_{UU}(T) = 0$.

$$I_{LL} + \lambda I_{UU} = \frac{\alpha_4 + \lambda \alpha_1}{(c + \lambda \theta_{LL})} \left(1 - e^{(c + \lambda \theta_{LL})(T - t)} \right)$$

For the two values of λ_3 and λ_4 , we obtain two simultaneous equations:

$$I_{LL} + \lambda_3 I_{UU} = C_1 \left(1 - e^{D_1 (T-t)} \right)$$
(A2)

$$I_{LL} + \lambda_4 I_{UU} = C_2 \left(1 - e^{D_2(T-t)} \right)$$
 (A3)

where $C_1 = \frac{\alpha_4 + \lambda_3 \alpha_1}{(c + \lambda_3 \theta_{LL})}$, $D_1 = (c + \lambda_3 \theta_{LL})$, $C_2 = \frac{\alpha_4 + \lambda_4 \alpha_1}{(c + \lambda_4 \theta_{LL})}$, and $D_2 = (c + \lambda_4 \theta_{LL})$. Solving Equations (A2) and (A3), we obtain

$$\begin{cases} I_{LL}(t) = \frac{\lambda_4 C_1 \left(1 - e^{D_1(T-t)}\right) - \lambda_3 C_2 \left(1 - e^{D_2(T-t)}\right)}{\lambda_4 - \lambda_3} \\ I_{UU}(t) = \frac{C_1 \left(1 - e^{D_1(T-t)}\right) - C_2 \left(1 - e^{D_2(T-t)}\right)}{\lambda_3 - \lambda_4} \\ 0 \le t \le T \end{cases}$$
(A4)

Using the initial conditions $I_{LL}(0) = Q_{LL}$ and $I_{UU}(0) = Q_{UU}$ in Equation (A4), we obtain

$$\begin{cases} Q_{LL} = \frac{\lambda_4 C_1 (1 - e^{D_1 T_1}) - \lambda_3 C_2 (1 - e^{D_2 T_1})}{\lambda_4 - \lambda_3} \\ Q_{UU} = \frac{C_1 (1 - e^{D_1 T_1}) - C_2 (1 - e^{D_2 T_1})}{\lambda_3 - \lambda_4} \end{cases}$$
(A5)

Similarly, solving System (45) by Lagrange's multiplier method

$$I_{LU} + \mu I_{UL} = \frac{\alpha_3 + \mu \alpha_2}{(c + \mu \theta_{LU})} \left(1 - e^{-(c + \mu \theta_{LU})t} \right) \text{ where } \mu = \frac{(\theta_{UL} + \mu c)}{(c + \mu \theta_{LU})}$$

So, for μ_3 and μ_4 , we obtained the solutions as

$$\begin{cases} I_{LU}(t) = \frac{\mu_4 C_3 (1 - e^{D_3(T-t)}) - \mu_3 C_4 (1 - e^{D_4(T-t)})}{\mu_4 - \mu_3} \\ I_{UL}(t) = \frac{C_3 (1 - e^{D_3(T-t)}) - C_4 (1 - e^{D_4(T-t)})}{\mu_3 - \mu_4} \\ 0 \le t \le T \end{cases}$$
(A6)

where $C_3 = \frac{\alpha_3 + \mu_3 \alpha_2}{(c + \mu_3 \theta_{LU})}$, $D_3 = (c + \mu_3 \theta_{LU})$, $C_4 = \frac{\alpha_3 + \mu_4 \alpha_2}{(c + \mu_4 \theta_{LU})}$, and $D_4 = (c + \mu_4 \theta_{LU})$. Using the initial conditions $I_{LU}(0) = Q_{LU}$ and $I_{UL}(0) = Q_{UL}$ in Equation (A6)

Using the initial conditions $I_{LU}(0) = Q_{LU}$ and $I_{UL}(0) = Q_{UL}$ in Equation (A6) we obtain $\int Q_{UL} = \mu_4 C_3 (1 - e^{D_3 T}) - \mu_3 C_4 (1 - e^{D_4 T})$

$$\begin{cases} Q_{LU} = \frac{\mu_4 C_3 (1 - e^{D_3 T}) - \mu_3 C_4 (1 - e^{D_4 T})}{\mu_4 - \mu_3} \\ Q_{UL} = \frac{C_3 (1 - e^{D_3 T}) - C_4 (1 - e^{D_4 T})}{\mu_3 - \mu_4} \end{cases}$$

Several relevant costs and the earned revenue will be obtained as follows:

- (i) The replenishment cost is a constant, K, as in case 1.
- (ii) Holding cost: Let $hc = [[hc_{LL}, hc_{LU}], [hc_{UL}, hc_{UU}]]$ be the per unit holding cost per unit of time. Then, the holding cost $HC_2 = [[HC_{LL}, HC_{LU}], [HC_{UL}, HC_{UU}]]$ is given by

$$HC_{LL} = hc_{LL} \int_{0}^{T} I_{LL}(t) dt = hc_{LL} \int_{0}^{T} \frac{\lambda_4 C_1 \left(1 - e^{D_1(T-t)}\right) - \lambda_3 C_2 \left(1 - e^{D_2(T-t)}\right)}{\lambda_4 - \lambda_3} dt$$
$$= \frac{hc_{LL}}{\lambda_4 - \lambda_3} \left[\frac{C_1 \lambda_4}{D_1} \left(D_1 T + 1 - e^{D_1 T} \right) - \frac{C_2 \lambda_3}{D_2} \left(D_2 T + 1 - e^{D_2 T} \right) \right]$$

$$\begin{aligned} HC_{LU} &= hc_{LU} \int_{0}^{T} I_{LU}(t) dt = hc_{LU} \int_{0}^{T} \frac{M_{3}\mu_{2} \left(1 - e^{N_{3}(T-t)}\right) - M_{4}\mu_{1} \left(1 - e^{N_{4}(T-t)}\right)}{\mu_{2} - \mu_{1}} dt \\ &= \frac{hc_{LU}}{\mu_{4} - \mu_{3}} \left[\frac{C_{3}\mu_{4}}{D_{3}} \left(D_{3}T + 1 - e^{D_{3}T}\right) - \frac{C_{4}\mu_{3}}{D_{4}} \left(D_{4}T + 1 - e^{D_{4}T}\right) \right] \\ HC_{UL} &= hc_{UL} \int_{0}^{T} I_{UL}(t) dt = hc_{UL} \int_{0}^{T} \frac{C_{3} \left(1 - e^{D_{3}(T-t)}\right) - C_{4} \left(1 - e^{D_{4}(T-t)}\right)}{\mu_{3} - \mu_{4}} dt \\ &= \frac{hc_{UL}}{\mu_{3} - \mu_{4}} \left[\frac{C_{3}}{D_{3}} \left(D_{3}T + 1 - e^{D_{3}T}\right) - \frac{C_{4}}{D_{4}} \left(D_{4}T + 1 - e^{D_{4}T}\right) \right] \end{aligned}$$

$$HC_{UU} = hc_{UU} \int_{0}^{T} I_{UU}(t) dt = hc_{UU} \int_{0}^{T} \frac{C_1 \left(1 - e^{D_1(T-t)}\right) - C_2 \left(1 - e^{D_2(T-t)}\right)}{\lambda_3 - \lambda_4} dt$$

$$= \frac{hc_{UU}}{\lambda_3 - \lambda_4} \left[\frac{C_1}{D_1} \left(D_1 T + 1 - e^{D_1 T} \right) - \frac{C_2}{D_2} \left(D_2 T + 1 - e^{D_2 T} \right) \right]$$

(iii) Purchase cost: Let $pc = [[pc_{LL}, pc_{LU}], [pc_{UL}, pc_{UU}]]$ be the purchase cost per unit of time. Then, the purchasing cost $PC_2 = [[PC_{LL}, PC_{LU}], [PC_{UL}, PC_{UU}]]$ is given by

$$PC_{LL} = pc_{LL}Q_{LL} = pc_{LL}\frac{\lambda_4 C_1 \left(1 - e^{D_1 T}\right) - \lambda_3 C_2 \left(1 - e^{D_2 T}\right)}{\lambda_4 - \lambda_3}$$
$$PC_{LU} = pc_{LU}Q_{LU} = pc_{LU}\frac{\mu_4 C_3 \left(1 - e^{D_3 T}\right) - \mu_3 C_4 \left(1 - e^{D_4 T}\right)}{\mu_4 - \mu_3}$$
$$PC_{UL} = pc_{UL}Q_{UL} = pc_{UL}\frac{C_3 \left(1 - e^{D_3 T}\right) - C_4 \left(1 - e^{D_4 T}\right)}{\mu_3 - \mu_4}$$
$$PC_{UU} = pc_{UU}Q_{UU} = pc_{UU}\frac{C_1 \left(1 - e^{D_1 T}\right) - C_2 \left(1 - e^{D_2 T}\right)}{\lambda_3 - \lambda_4}$$

(iv) The total sales revenue is $SR_2 = [[SR_{LL}, SR_{LU}], [SR_{UL}, SR_{UU}]]$ during the entire cycle. Then,

$$SR_{LL} = p_{LL} \int_{0}^{T} \{a - bp_{UU} + cI_{LL}(t)\}dt$$

$$= p_{LL}(a - bp_{UU})T + \frac{p_{LL}c}{\lambda_4 - \lambda_3} \left[\frac{C_1\lambda_4}{D_1} \left(D_1T + 1 - e^{D_1T} \right) - \frac{C_2\lambda_3}{D_2} \left(D_2T + 1 - e^{D_2T} \right) \right]$$
$$SR_{LU} = p_{LU} \int_{0}^{T} \{a - bp_{UL} + cI_{LU}(t)\} dt$$

$$= p_{LU}(a - bp_{UL})T + \frac{p_{LU}c}{\mu_4 - \mu_3} \left[\frac{C_3\mu_4}{D_3} \left(D_3T + 1 - e^{D_3T} \right) - \frac{C_4\mu_3}{D_4} \left(D_4T + 1 - e^{D_4T} \right) \right]$$
$$SR_{UL} = p_{UL} \int_0^T \{a - bp_{LU} + cI_{UL}(t)\} dt$$

$$= p_{UL}(a - b_{LU})T + \frac{p_{UL}c}{\mu_3 - \mu_4} \left[\frac{C_3}{D_3} \left(D_3 T + 1 - e^{D_3 T} \right) - \frac{C_4}{D_4} \left(D_4 T + 1 - e^{D_4 T} \right) \right]$$
$$SR_{UU} = p_{UU} \int_0^T \{a - bp_{LL} + cI_{UU}(t)\} dt$$

$$= p_{UU}(a - bp_{LL})T + \frac{p_{UU}c}{\lambda_3 - \lambda_4} \left[\frac{C_1}{D_1} \left(D_1T + 1 - e^{D_1T} \right) - \frac{C_2}{D_2} \left(D_2T + 1 - e^{D_2T} \right) \right]$$

References

- 1. Moore, R.E.; Kearfott, R.B.; Cloud, M.J. *Introduction to Interval Analysis*; Society for Industrial and Applied Mathematics: Philadelphia, PA, USA, 2009.
- 2. Aubin, J.P.; Frankowska, H. Set-Valued Analysis; Birkhäuser: Boston, MA, USA, 2009.

- 3. Lakshmikantham, V.; Bhaskar, T.G.; Devi, J.V. *Theory of Set Differential Equations in Metric Spaces*; Cambridge Scientific Publishers Ltd.: Cambridge, UK, 2006.
- 4. Candau, Y.; Raissi, T.; Ramdani, N.; Ibos, L. Complex interval arithmetic using polar form. *Reliab. Comput.* **2006**, *12*, 1–20. [CrossRef]
- Ramos, C.C.; Martins, N.; Pinto, P.R. Interval maps from Cuntz-Krieger algebras. J. Math. Anal. Appl. 2011, 374, 347–354. [CrossRef]
- 6. Garg, H.; Rani, M.; Sharma, S.P.; Vishwakarma, Y. Intuitionistic fuzzy optimization technique for solving multi-objective reliability optimization problems in interval environment. *Expert Syst. Appl.* **2014**, *41*, 3157–3167. [CrossRef]
- Wang, W.; Xiong, J.; Xie, M. A study of interval analysis for cold-standby system reliability optimization under parameter uncertainty. *Comput. Ind. Eng.* 2016, 97, 93–100. [CrossRef]
- 8. Hladík, M. Complexity of necessary efficiency in interval linear programming and multi-objective linear programming. *Optim. Lett.* **2012**, *6*, 893–899. [CrossRef]
- 9. Jiang, C.; Han, X.; Liu, G.; Liu, G. A non-linear interval number programming method for uncertain optimization problems. *Eur. J. Oper. Res.* **2008**, *188*, 1–13. [CrossRef]
- 10. Žilinskas, J.; Bogle, I.D.L. A survey of methods for the estimation ranges of functions using interval arithmetic. *Models Algorithms Glob. Optim.* **2007**, *4*, 97–108.
- 11. Popova, E.D.; Hladik, M. Outer enclosures to the parametric AE solution set. Soft Comput. 2013, 17, 1403–1414. [CrossRef]
- 12. Sahoo, L.; Bhunia, A.K.; Kapur, P.K. Genetic algorithm based multi-objective reliability optimization in interval environment. *Comput. Ind. Eng.* **2012**, *62*, 152–160. [CrossRef]
- 13. Sahoo, L.; Bhunia, A.K.; Roy, D. Reliability optimization with and low-level redundancies in interval environment via genetic algorithm. *Int. J. Syst. Assur. Manag.* 2014, *5*, 513–523. [CrossRef]
- 14. Pal, D.; Mahapatra, G.S.; Samanta, G.P. Optimal harvesting of pre-predator system with interval biological parameters: A biological model. *Math. Biosci.* 2013, 241, 181–187. [CrossRef] [PubMed]
- 15. Rahman, M.S.; Shaikh, A.A.; Bhunia, A.K. On the space of Type-2 interval with limit, continuity and differentiability of Type-2 interval-valued functions. *arXiv* **2019**, arXiv:1907.00644.
- Rahman, M.S.; Shaikh, A.A.; Bhunia, A.K. On Type-2 interval with interval mathematics and order relations: Its applications in inventory control. *Int. J. Syst. Sci. Oper. Logist.* 2021, *8*, 283–295. [CrossRef]
- Rahman, M.S.; Shaikh, A.A.; Ali, I.; Bhunia, A.K.; Fügenschuh, A. A theoretical framework for optimality conditions of non-linear type-2 interval-valued unconstrained and constrained optimization problems using type-2 interval order relations. *Mathematics* 2021, 9, 908. [CrossRef]
- Rahman, M.S.; Duary, A.; Shaikh, A.A.; Bhunia, A.K. An application of real coded Self-organizing Migrating Genetic Algorithm on a two-warehouse inventory problem with Type-2 interval valued inventory costs via mean bounds optimization technique. *Appl. Soft Comput.* 2022, 124, 109085. [CrossRef]
- Das, S.; Rahman, M.S.; Mahato, S.K.; Shaikh, A.A.; Bhunia, A.K. On the Parametric Representation of Type-2 Interval and Ranking: Its Application in the Unconstrained Non-linear Programming Problem Under Type-2 Interval Uncertainty. In *Computational Modelling in Industry 4.0: A Sustainable Resource Management Perspective*; Springer Nature: Singapore, 2022; pp. 281–297.
- Rahaman, M.; Mondal, S.P.; Alam, S.; Metwally, A.S.M.; Salahshour, S.; Salimi, M.; Ahmadian, A. Manifestation of interval uncertainties for fractional differential equations under conformable derivative. *Chaos Solitons Fractals* 2022, 165, 112751. [CrossRef]
- 21. Stefanini, L.; Bede, B. Generalized Hukuhara differentiability of interval-valued functions and interval differential equations. *Non-Linear Anal. Theory Methods Appl.* **2009**, *71*, 1311–1328. [CrossRef]
- Bede, B.; Stefanini, L. Numerical Solution of Interval Differential Equations with Generalized Hukuhara Differentiability. In Proceedings of the Joint 2009 International Fuzzy Systems Association World Congress and 2009 European Society of Fuzzy Logic and Technology Conference, Lisbon, Portugal, 20–24 July 2009; pp. 730–735.
- 23. Stefanini, L. A generalization of Hukuhara difference and division for interval and fuzzy arithmetic. *Fuzzy Sets Syst.* 2010, 161, 1564–1584. [CrossRef]
- 24. Tao, J.; Zhang, Z. Properties of interval-valued function space under the gH-difference and their application to semi-linear interval differential equations. *Adv. Diff. Equ.* **2016**, 2016, 45. [CrossRef]
- Lupulescu, V. Hukuhara differentiability of interval-valued functions and interval differential equations on time scales. *Inf. Sci.* 2013, 248, 50–67. [CrossRef]
- 26. Agarwal, R.P.; Lakshmikantham, V.; Nieto, J.J. On the concept of solution for fractional differential equations with uncertainty. *Non-Linear Anal. Theory Methods Appl.* **2010**, *72*, 2859–2862. [CrossRef]
- 27. Allahviranloo, T.; Ahmadi, M.B. Fuzzy Laplace transforms. Soft Comput. 2010, 14, 235–243. [CrossRef]
- Salahshour, S.; Allahviranloo, T.; Abbasbandy, S. Solving fuzzy fractional differential equations by fuzzy Laplace transforms. *Commun. Non-Linear Sci. Numer. Simul.* 2012, 17, 1372–1381. [CrossRef]
- 29. Salahshour, S.; Allahviranloo, T. Applications of fuzzy Laplace transforms. Soft Comput. 2013, 17, 145–158. [CrossRef]
- Ahmad, M.Z.; Hasan, M.K.; Abbasbandy, S. Solving fuzzy fractional differential equations using Zadeh's extension principle. *Sci.* World J. 2013, 2013, 454969. [CrossRef]

- 31. Van-Ngo, H.; Lupulescu, V.; O'Regan, D. A note on initial value problems for fractional fuzzy differential equations. *Fuzzy Sets Syst.* **2018**, *347*, 54–69.
- 32. Rahaman, M.; Mondal, S.P.; Alam, S.; Goswami, A. Synergetic study of inventory management problem in uncertain environment based on memory and learning effects. *Sadhana* **2021**, *46*, 39. [CrossRef]
- Rahaman, M.; Mondal, S.P.; Alam, S. An estimation of effects of memory and learning experience on the EOQ model with price dependent demand. *RAIRO Oper. Res.* 2021, 55, 2991–3020. [CrossRef]
- 34. Lupulescu, V. Fractional calculus for interval-valued functions. Fuzzy Sets Syst. 2015, 265, 63–85. [CrossRef]
- 35. Salahshour, S.; Ahmadian, A.; Ismail, F.; Baleanu, D.; Senu, N. A new fractional derivative for differential equation of fractional order under interval uncertainty. *Adv. Mech. Eng.* **2015**, *7*, 1687814015619138. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.